Conditional decisions under objective and subjective ambiguity in Dempster-Shafer theory

Davide Petturiti^{a,*}, Barbara Vantaggi^b

^aDip. Economia, Università di Perugia, 06100 Perugia, Italy ^bDip. MEMOTEF, "La Sapienza" Università di Roma, 00185 Roma, Italy

Abstract

This paper deals with conditional decisions on generalized Anscombe-Aumann acts mapping states of the world to finitely additive probabilities on the set of menus of consequences, the latter conveying a form of "objective" ambiguity. If the decision maker has a systematic pessimistic/optimistic attitude towards "objective" ambiguity, acts reduce to functions mapping states of the world to belief/plausibility functions on consequences. We provide a system of axioms assuring the representability of a family of conditional preference relations on such acts by a conditional functional in which "subjective" uncertainty is modeled through a conditional belief/plausibility function on the states of the world, obeying to a suitable axiomatic definition.

Keywords: Belief functions, Finitely additive probabilities, Ambiguity, Conditional preferences

1. Introduction

Choices based on subjective probabilities find their roots in the seminal works by Savage [41] and Anscombe-Aumann [2]: such models, though relying on different decision settings, assume additivity in the quantification of decision maker's uncertainty. In particular, in the classical formulation proposed by Anscombe-Aumann we distinguish between "objective" uncertainty related to consequences and "subjective" uncertainty related to states

Preprint submitted to Fuzzy Sets and Systems https://doi.org/10.1016/j.fss.2022.02.011

^{*}Corresponding author.

Email addresses: davide.petturiti@unipg.it (Davide Petturiti), barbara.vantaggi@uniroma1.it (Barbara Vantaggi)

of the world: the first is exogenously quantified, while the second is encoded in the decision maker's preferences.

In many circumstances, due to partial knowledge, uncertainty cannot be expressed by a single probability measure, but we need to manage a class of probability measures: this configures a situation of *ambiguity*.

We consider generalized Anscombe-Aumann acts mapping states of the world to finitely additive probabilities on the set of menus of consequences [33]. Each of such finitely additive probabilities encodes a form of "objective" ambiguity as it induces a closed convex set of finitely additive probabilities on the set of consequences whose lower/upper envelope is a belief/plausibility function in the Dempster-Shafer theory [9, 44]. Hence, we interpret such finitely additive probabilities as generalized lotteries since they allow to model partially known randomizing devices (like an urn or a roulette wheel), that result in a class of finitely additive probabilities whose lower/upper envelope is a belief/plausibility function, like in the well-known Ellsberg's urn paradox [12]. The term "objective" ambiguity is borrowed from [48] (see also [1, 37]) where the author considers acts mapping states of the world to non-empty compact convex polyhedral sets of probabilities on consequences.

As acknowledged by the survey papers [13] and [21], most of the efforts in decision theory have been devoted to model "subjective" ambiguity, that is to ambiguity in "subjective" uncertainty evaluations. This research stream starts with the seminal papers by Gilboa and Schmeidler, dealing with a capacity [43] or a set of probability measures [22], respectively. For this, we assume that also "subjective" uncertainty can be ambiguous, still referring to the Demspter-Shafer framework. Further, in order to deal with dynamic decision problems, a suitable notion of conditioning for belief/plausibility functions is necessary.

Starting from Dempster's original work [9], many researches have tried to propose a suitable notion of conditioning for belief/plausibility functions and, more generally, for non-additive uncertainty measures. Conditioning has been approached essentially proposing *conditiong rules* either working directly with a non-additive measure or working with the set of probability measures dominating/dominated by it. In particular, for rules of the first group we refer to [5, 7, 23, 27] and for those of the second group to [31, 49]. For a discussion on both approaches to conditioning we cite, for instance, [28, 32]. Here, we consider axiomatic definitions of conditional belief and plausibility functions that allow to define the conditional measure of every conditional event E|H with $H \neq \emptyset$ (see [6]). Such axiomatic definitions generalize the so-called *Dempster rule* [9] and *product rule* [45, 46] and have conditional probability in the sense of Dubins [11] as a particular case. These axiomatic definitions make sense even if H is an event of null measure, as a pandemic or a terror attack. We stress that, as shown in [34], "null" events play a crucial role in the analysis of a game (see also [36]), thus this axiomatic approach seems to be the most suitable to model "subjective" uncertainty.

We consider a conditional decision model involving the above generalization of Anscombe-Aumann acts, assuming that the decision maker is able to provide a family of preference relations on acts indexed by the set of non-impossible events. Every preference relation can be interpreted as comparing acts under a particular hypothesis (or "scenario").

We provide a system of axioms assuring the representability of the family of conditional preference relations through a conditional functional parametrized by a bounded utility function on consequences and a full conditional belief function on the states. Such functional turns out to be a "subjective" conditional Choquet expectation of a state-contingent "objective" mixture of convex combinations of infima and suprema of utility on menus. Thus, the conditional functional involves a double integration, where the inner integral is inspired to the functional proposed by [17] in a Savage's setting, through acts that map states of the world to non-empty sets of consequences. The obtained conditional functional is then specialized to a "subjective" conditional Choquet expectation of a state-contingent "objective" lower/upper expected utility, the latter reducing to a state-contingent Choquet expected utility. The last two conditional functionals generalize to arbitrary sets of consequences and states of the world (besides introducing "subjective" ambiguity), those introduced in [40] that, in turn, generalize the conditional version of the Anscombe-Aumann model given in [36] by introducing "objective" ambiguity. Then, we consider analogous models in which the external full conditional belief function is replaced by a full conditional plausibility function. In the end, we consider the special case of a Bayesian decision maker, whose "subjective" uncertainty is unambiguous and expressed by a conditional probability in the sense of Dubins [11]. In this case, the preference relations are shown to satisfy a strong form of *dynamic consistency* (for a discussion on dynamic consistency we refer, e.g., to [47]) that takes into account the possibility of conditioning on "null" events as in [36].

The paper is structured as follows. Section 2 presents a motivating decision problem for the models discussed in this paper. Section 3 deals with conditioning in Dempster-Shafer theory, while Section 4 introduces the decision-theoretic setting and the representation theorems. Section 5 shows that the preference pattern singled out in Section 2 is not consistent with a

Bayesian decision maker and that conditioning to a "null" event is crucial. Finally, Section 6 collects conclusions. Proofs of results are reported in Appendix A to improve readability.

2. A motivating decision problem

An agent in an Italian insurance company needs to decide between four investment strategies where the adopted financial instrument is contingent on the state of the world.

Consider the set of states of the world $S = \{s_1, s_2, s_3, s_4\}$, where:

- $s_1 =$ "At least one Covid-19 lockdown in Italy next year and Italian GDP increases next year",
- $s_2 =$ "At least one Covid-19 lockdown in Italy next year and Italian GDP does not increase next year",
- $s_3 =$ "No Covid-19 lockdown in Italy next year and Italian GDP increases next year",
- $s_4 =$ "No Covid-19 lockdown in Italy next year and Italian GDP does not increase next year".

Such states of the world give rise to the events:

- $C = \{s_1, s_2\} =$ "At least one Covid-19 lockdown in Italy next year",
- $C^c = \{s_3, s_4\} =$ "No Covid-19 lockdown in Italy next year".

The agent has at his/her disposal three different financial instruments maturing after one year. The result in one year of each financial instrument can be a loss of $\in 1000$, a null gain, or a gain of $\in 1000$, i.e., instruments take values in $X = \{-\in 1000, \in 0, \in 1000\}$.

The usual decision-theoretic approach would be to identify each instrument with a probability distribution on X. Unfortunately, due to partial knowledge on previous performances of instruments, the agent has only the following information:

- **Instrument 1:** it is only known that it results in a null gain in 50% of cases.
- Instrument 2: it is only known that it results in a gain of €1000 in 30% of cases.

Instrument 3: it is only known that it results in a loss of €1000 in 30% of cases.

We have that instrument *i* determines a class of probability measures \mathbf{P}^i on $\mathcal{P}(X)$ compatible with the available information:

$$\begin{split} \mathbf{P}^{1} &= \{P \,:\, P \text{ is a probability measure on } \mathcal{P}(X), \gamma \in [0, 0.5], \\ &P(\{-1000\}) = \gamma, P(\{0\}) = 0.5, P(\{1000\}) = 0.5 - \gamma\}, \\ \mathbf{P}^{2} &= \{P \,:\, P \text{ is a probability measure on } \mathcal{P}(X), \gamma \in [0, 0.7], \\ &P(\{-1000\}) = \gamma, P(\{0\}) = 0.7 - \gamma, P(\{1000\}) = 0.3\}, \\ \mathbf{P}^{3} &= \{P \,:\, P \text{ is a probability measure on } \mathcal{P}(X), \gamma \in [0, 0.7], \\ &P(\{-1000\}) = 0.3, P(\{0\}) = \gamma, P(\{1000\}) = 0.7 - \gamma\}. \end{split}$$

For every i = 1, 2, 3, if we consider the lower envelope on the elements of $\mathcal{P}(X)$ defined as $\varphi_i = \min \mathbf{P}^i$, we get a non-additive uncertainty measure which reveals to be a belief function (whose definition and properties are recalled in Section 3). Denoting $x_1 = -1000$, $x_2 = 0$, $x_3 = 1000$, $A_i = \{x_i\}$ and $A_{ij} = \{x_i, x_j\}$, we have that

$\mathcal{P}(X)$	Ø	A_1	A_2	A_3	A_{12}	A_{13}	A_{23}	X
φ_1	0	0	0.5	0	0.5	0.5	0.5	1
φ_2	0	0	0	0.3	0.7	0.3	0.3	1
$arphi_3$	0	0.3	0	0	0.3	0.3	0.7	1

Now, consider the following four investment strategies in which the adopted financial instrument is contingent on the state of the world:

S	s_1	s_2	s_3	s_4
f	Instr. 2	Instr. 2	Instr. 2	Instr. 2
g	Instr. 1	Instr. 3	Instr. 3	Instr. 3
h	Instr. 3	Instr. 3	Instr. 2	Instr. 2
l	Instr. 3	Instr. 3	Instr. 3	Instr. 3

We assume that the agent needs to decide among the above investment strategies, immediately after the end of 2021 Covid-19 lockdown. This makes of particular importance to take into account the scenarios C and C^c .

Suppose the agent is *ex ante* indifferent between f, g and f, h, while strictly prefers l to f. Moreover, he/she strictly prefers g to f conditionally on both the scenarios C and C^c . In symbol, we write

$$f \sim_S g$$
, $f \sim_S h$, $f \prec_S l$, $f \prec_C g$, and $f \prec_{C^c} g$.

Further, suppose, as natural, that such an agent is a profit maximizer. We also assume that the agent is systematically pessimistic in coping with "objective" uncertainty, meaning that he/she always identifies each menu of consequences with the less preferred consequence in it.

Representing instrument i with the corresponding belief function φ_i on $\mathcal{P}(X)$, for i = 1, 2, 3, each investment strategy can be seen as a function mapping states of the world to belief functions on consequences. This configures a two-stage decision problem in which first the state is chosen by nature and then the consequence is chosen according to a partially known randomizing device acting in agreement with one of the belief functions above.

In [40], we proposed a model able to deal with a decision problem like the one presented so far. Such model assumes the agent is Bayesian, that is the agent uses conditional probability to encode his/her "subjective" beliefs on the states of the world.

As will be shown in Section 4, the preference pattern $f \prec_C g$, $f \prec_{C^c} g$ and $f \sim_S g$ violates the *strong dynamic consistency* axiom, that would require $f \prec_S g$, instead. Such a property must be satisfied by a family of conditional preference relations to be consistent with a Bayesian agent. Further, the preference statements $f \sim_S h$ and $f \prec_S l$ imply that event C is a "null" event, that is the agent finds C as unexpected. This evaluation of C is motivated by the current status of the vaccination campaign in several countries.

The decision problem above shows two different kinds of ambiguity: "objective" ambiguity on the consequences and "subjective" ambiguity on the states. This requires a departure from probability theory, in modeling both "objective" and "subjective" uncertainty. Moreover, the example shows a situation in which conditioning on a "null" event is crucial.

We maintain that Dempster-Shafer theory is the theory of uncertainty closest to probability and yet, under a suitable notion of conditioning, able to treat situations like those described above.

3. Conditioning in Dempster-Shafer theory

In this section we collect some preliminary material working on an abstract non-empty set Ω . Denote by $\mathcal{P}(\Omega)$ the power set of Ω and by $\mathcal{U} = \mathcal{P}(\Omega)^0 = \mathcal{P}(\Omega) \setminus \{\emptyset\}.$

A belief function (or completely monotone capacity) [9, 44] on $\mathcal{P}(\Omega)$ is a function $\varphi : \mathcal{P}(\Omega) \to [0, 1]$ satisfying the following properties:

(i) $\varphi(\emptyset) = 0$ and $\varphi(\Omega) = 1$;

(*ii*) for every $k \geq 2$ and every $A_1, \ldots, A_k \in \mathcal{P}(\Omega)$,

$$\varphi\left(\bigcup_{i=1}^k A_i\right) \geq \sum_{\emptyset \neq I \subseteq \{1,\dots,k\}} (-1)^{|I|+1} \varphi\left(\bigcap_{i \in I} A_i\right).$$

In particular, φ is a *finitely additive probability (measure)* if it satisfies the above inequality as an equality. The *dual* function ψ defined, for every $A \in \mathcal{P}(\Omega)$, as $\psi(A) = 1 - \varphi(A^c)$, is said *plausibility function* (or *completely alternating capacity*) and satisfies the above inequality in the opposite direction, switching intersections and unions.

Every belief function φ on $\mathcal{P}(\Omega)$ with dual ψ induces a non-empty closed convex set of finitely additive probabilities (see, e.g., [25, 43]) on $\mathcal{P}(\Omega)$:

 $\operatorname{core}(\varphi) = \{P : P \text{ is a finitely additive probability on } \mathcal{P}(\Omega), \varphi \leq P\},\$

such that $\varphi = \min \operatorname{core}(\varphi)$ and $\psi = \max \operatorname{core}(\varphi)$, where the minimum and maximum are pointwise on $\mathcal{P}(\Omega)$. As shown in [42], the Choquet integral of every bounded function $f : \Omega \to \mathbb{R}$ with respect to φ and ψ has a lower/upper expectation interpretation

$$\begin{split} \oint_{\Omega} f(\omega)\varphi(\mathrm{d}\omega) &= \min_{P\in\mathbf{core}(\varphi)} \int_{\Omega} f(\omega)P(\mathrm{d}\omega), \\ \oint_{\Omega} f(\omega)\psi(\mathrm{d}\omega) &= \max_{P\in\mathbf{core}(\varphi)} \int_{\Omega} f(\omega)P(\mathrm{d}\omega). \end{split}$$

In all the paper, integrals with respect to a finitely additive probability are of Stieltjes type [3].

Theorem A in [24] shows that every belief function φ on $\mathcal{P}(\Omega)$ corresponds to a unique finitely additive probability measure μ_{φ} defined on an algebra \mathfrak{A} (possibly strictly contained in $\mathcal{P}(\mathcal{U})$) such that, for every $A \in \mathcal{P}(\Omega)$

$$\varphi(A) = \int_{\mathcal{U}} \nu_B(A) \mu_{\varphi}(\mathrm{d}B),\tag{1}$$

where $\nu_B : \mathcal{P}(\Omega) \to [0,1]$ is a vacuous belief function (or unanimity game) on B such that $\nu_B(A) = 1$ if $B \subseteq A$ and 0 otherwise. In analogy with the case of a finite Ω [4, 25], we refer to the finitely additive probability μ_{φ} on \mathfrak{A} as the *Möbius inverse* of φ and \mathfrak{A} is the algebra on \mathcal{U} generated by $\{\tilde{A} : A \in \mathcal{U}\}$ where, for every $A \in \mathcal{P}(\Omega)$,

$$\tilde{A} = \{ B \in \mathcal{U} : B \subseteq A \},\$$

In case of a finite Ω we have that $\mathfrak{A} = \mathcal{P}(\mathcal{U})$.

Notice, that μ_{φ} can be extended, generally not in a unique way, to a finitely additive probability defined on the whole $\mathcal{P}(\mathcal{U})$.

Conversely, every finitely additive probability measure μ on $\mathcal{P}(\mathcal{U})$ determines a belief function φ_{μ} on $\mathcal{P}(\Omega)$ but, in general, there can be more finitely additive probabilities giving rise to the same belief function. The main feature of finitely additive probabilities on $\mathcal{P}(\mathcal{U})$ is that only their restriction to \mathfrak{A} is meaningful in order to get a representation of the corresponding belief function.

With a dual argument we have that every finitely additive probability measure μ on $\mathcal{P}(\mathcal{U})$ determines also a plausibility function ψ_{μ} on $\mathcal{P}(\Omega)$.

A finitely additive probability μ defined on $\mathcal{P}(\mathcal{U})$ allows to have an additive expression of the Choquet integral of a bounded function $f: \Omega \to \mathbb{R}$ with respect to φ_{μ} and ψ_{μ} . Indeed, by Corollary in [35] (see also Proposition 6 in [17]) we have

$$\begin{split} \oint_{\Omega} f(\omega) \varphi_{\mu}(\mathrm{d}\omega) &= \int_{\mathcal{U}} \left(\inf_{\omega \in B} f(\omega) \right) \mu(\mathrm{d}B), \\ \oint_{\Omega} f(\omega) \psi_{\mu}(\mathrm{d}\omega) &= \int_{\mathcal{U}} \left(\sup_{\omega \in B} f(\omega) \right) \mu(\mathrm{d}B). \end{split}$$

Starting from the seminal work by Dempster [9] (see also [14, 30, 50]), conditioning has been introduced either by providing a conditioning rule "directly" for non-additive uncertainty measures, or applying the conditioning to every probability in the core and then defining the conditional non-additive measure as their envelope.

In the decision theory literature both alternatives have been considered, in particular for the first group we refer to [5, 8, 23, 27] and for the second group to [31, 49]. For a discussion on both approaches to conditioning we cite, for instance, [28, 32].

Given a belief function $\varphi : \mathcal{P}(\Omega) \to [0, 1]$ and its dual plausibility function ψ , the most popular conditioning rules are, for every $E|H \in \mathcal{P}(\Omega) \times \mathcal{P}(\Omega)^0$:

(Dempster rule [9]) provided $\psi(H) > 0$,

$$\varphi_D(E|H) = 1 - \psi_D(E^c|H) \text{ and } \psi_D(E|H) = \frac{\psi(E \cap H)}{\psi(H)};$$

(Product rule [45, 46]) provided $\varphi(H) > 0$,

$$\varphi_P(E|H) = \frac{\varphi(E \cap H)}{\varphi(H)}$$
 and $\psi_P(E|H) = 1 - \varphi_P(E^c|H);$

(Bayesian rule [31, 50]) provided $\varphi(E \cap H) + \psi(E^c \cap H) > 0$,

$$\varphi_B(E|H) = \frac{\varphi(E \cap H)}{\varphi(E \cap H) + \psi(E^c \cap H)}$$
 and $\psi_B(E|H) = 1 - \varphi_B(E^c|H)$.

This paper adopts the conditioning rule expressed by the axiomatic definition of conditional belief and plausibility functions generalizing the product and Dempster rule, respectively. For a discussion on the different axiomatic definitions see [6] (see also [39]).

Definition 1. A function $\varphi : \mathcal{P}(\Omega) \times \mathcal{P}(\Omega)^0 \to [0,1]$ is a full conditional belief function on $\mathcal{P}(\Omega)$ if it satisfies the following conditions:

- (i) $\varphi(E|H) = \varphi(E \cap H|H)$, for every $E \in \mathcal{P}(\Omega)$ and $H \in \mathcal{P}(\Omega)^0$;
- (ii) $\varphi(\cdot|H)$ is a belief function on $\mathcal{P}(\Omega)$, for every $H \in \mathcal{P}(\Omega)^0$;
- (iii) $\varphi(E \cap F|H) = \varphi(E|H) \cdot \varphi(F|E \cap H)$, for every $H, E \cap H \in \mathcal{P}(\Omega)^0$ and $E, F \in \mathcal{P}(\Omega)$.

Definition 2. A function $\psi : \mathcal{P}(\Omega) \times \mathcal{P}(\Omega)^0 \to [0,1]$ is a full conditional plausibility function on $\mathcal{P}(\Omega)$ if it satisfies the following conditions:

- (i) $\psi(E|H) = \psi(E \cap H|H)$, for every $E \in \mathcal{P}(\Omega)$ and $H \in \mathcal{P}(\Omega)^0$;
- (ii) $\psi(\cdot|H)$ is a plausibility function on $\mathcal{P}(\Omega)$, for every $H \in \mathcal{P}(\Omega)^0$;
- (iii) $\psi(E \cap F|H) = \psi(E|H) \cdot \psi(F|E \cap H)$, for every $H, E \cap H \in \mathcal{P}(\Omega)^0$ and $E, F \in \mathcal{P}(\Omega)$.

Notice that condition (iii) in Definition 1 can be replaced by the chain rule condition (iii')

$$\varphi(A|C) = \varphi(A|B) \cdot \varphi(B|C),$$

for every $A \in \mathcal{P}(\Omega)$ and $B, C \in \mathcal{P}(\Omega)^0$, with $A \subseteq B \subseteq C$. An analogous chain rule condition can replace condition *(iii)* in Definition 2. Also, notice that condition *(iii)* in Definition 1 implies, in particular, that, for every $E|H \in \mathcal{P}(\Omega) \times \mathcal{P}(\Omega)^0$ with $\varphi(H|\Omega) > 0$

$$\varphi(E|H) = \frac{\varphi(E \cap H|\Omega)}{\varphi(H|\Omega)}$$

An analogous property holds for $\psi(\cdot|\cdot)$. As usual, we can identify $\varphi(\cdot|\Omega)$ with the unconditional belief function $\varphi(\cdot)$ and $\psi(\cdot|\Omega)$ with the unconditional plausibility function $\psi(\cdot)$.

Let us stress that the conditional measures $\varphi(\cdot|\cdot)$ and $\psi(\cdot|\cdot)$ defined above are generally not *dual* (see [39]), as it can be $\varphi(E|H) \neq 1 - \psi(E^c|H)$, for some $E \in \mathcal{P}(\Omega)$ and $H \in \mathcal{P}(\Omega)^0$ with $H \neq \Omega$. The lack of preservation of duality is already known for the classical Dempster and product rules: as shown in the following example, in this context, even more degrees of freedom are introduced in case of conditioning on events of null measure.

In particular, a full conditional probability $P(\cdot|\cdot)$ on $\mathcal{P}(\Omega)$ in the sense of Dubins [11] is both a full conditional belief function and a full conditional plausibility function on $\mathcal{P}(\Omega)$ according to Definitions 1 and 2.

Example 1. Let $\Omega = \{\omega_1, \omega_2, \omega_3\}$ and consider the full conditional belief function on $\mathcal{P}(\Omega)$ given below

$\mathcal{P}(\Omega)$	Ø	$\{\omega_1\}$	$\{\omega_2\}$	$\{\omega_3\}$	$\{\omega_1,\omega_2\}$	$\{\omega_1,\omega_3\}$	$\{\omega_2,\omega_3\}$	Ω
$arphi(\cdot \Omega)$	0	0	0	0	0	0.5	0	1
$\varphi(\cdot \{\omega_1,\omega_2\})$	0	0.3	0.7	0	1	0.3	0.7	1
$\varphi(\cdot \{\omega_1,\omega_3\})$	0	0	0	0	0	1	0	1
$\varphi(\cdot \{\omega_2,\omega_3\})$	0	0	0.7	0.3	0.7	0.3	1	1
$\varphi(\cdot \{\omega_1\})$	0	1	0	0	1	1	0	1
$\varphi(\cdot \{\omega_2\})$	0	0	1	0	1	0	1	1
$\varphi(\cdot \{\omega_3\})$	0	0	0	1	0	1	1	1

The function $\varphi(\cdot|\cdot)$ is easily verified to agree with Definition 1. In particular, $\varphi(\cdot|\{\omega_1, \omega_2\})$ is well-defined, even though $\varphi(\{\omega_1, \omega_2\}|\Omega) = 0$.

Now, setting $\psi(E|\Omega) = 1 - \varphi(E^c|\Omega)$ for every $E \in \mathcal{P}(\Omega)$, we get a plausibility function $\psi(\cdot|\Omega)$ on $\mathcal{P}(\Omega)$ which is strictly positive on $\mathcal{P}(\Omega)^0$. Therefore, defining, for every $E|H \in \mathcal{P}(\Omega) \times \mathcal{P}(\Omega)^0$

$$\psi(E|H) = \frac{\psi(E \cap H|\Omega)}{\psi(H|\Omega)},$$

we get a full conditional plausibility on $\mathcal{P}(\Omega)$ agreeing with Definition 2. Note that $\varphi(\cdot|\Omega)$ and $\psi(\cdot|\Omega)$ are dual, while we have that

$$\varphi(\{\omega_1\}|\{\omega_1,\omega_2\}) = 0.3 \neq 0.5 = 1 - \psi(\{\omega_1\}^c | \{\omega_1,\omega_2\}).$$

This shows that $\varphi(\cdot|\cdot)$ and $\psi(\cdot|\cdot)$ are not dual in general.

4. Main model description

Consider the following decision-theoretic setting:

- X, an arbitrary non-empty set of consequences;
- $\mathcal{U} = \mathcal{P}(X) \setminus \{\emptyset\}$, the set menus, i.e., non-empty sets of consequences;
- M(U) = {μ : P(U) → [0,1]}, the set of all finitely additive probability measures defined on P(U);
- S, an arbitrary non-empty set of states of the world;
- $\mathcal{P}(S)$, the set of all events;
- $\mathcal{P}(S)^0 = \mathcal{P}(S) \setminus \{\emptyset\}$, the set of non-impossible events (i.e., scenarios or hypotheses);
- $\mathcal{F}_{\text{const}} = \{ \overline{\mu} \in \mathbf{M}(\mathcal{U})^S : \mu \in \mathbf{M}(\mathcal{U}) \}$, the set of constant acts, where $\overline{\mu} \in \mathcal{F}_{\text{const}}$ is such that $\overline{\mu}(s) = \mu$ for all $s \in S$;
- \mathcal{F}_{simple} , the set of simple acts (functions in $\mathbf{M}(\mathcal{U})^S$ with finite image);
- \mathcal{F} , a set of acts closed under pointwise convex combination, with $\mathcal{F}_{simple} \subseteq \mathcal{F} \subseteq \mathbf{M}(\mathcal{U})^S$: this implies $\mathcal{F}_{const} \subseteq \mathcal{F}$.

This decision-theoretic setting uses power sets in order to have a lighter presentation. The entire formulation of the problem can be restated using algebras of sets, provided to take care of measurability restrictions and some structural assumptions.

Consider a family $\{ \preceq_H \}_{H \in \mathcal{P}(S)^0}$ of preference relations on \mathcal{F} , indexed by the set of non-impossible events $\mathcal{P}(S)^0$. As usual \preceq_H is said to be a *weak order* if it is complete and transitive.

For every $H \in \mathcal{P}(S)^0$, we denote by \prec_H and \sim_H the asymmetric and symmetric parts of \preceq_H . Moreover, for every $f, g \in \mathcal{F}$, $f \preceq_H g$ means "f is not preferred to g under the hypothesis H", $f \prec_H g$ means "g is preferred to f under the hypothesis H", and $f \sim_H g$ means "f is indifferent to g under the hypothesis H".

Notice that the set $\mathbf{M}(\mathcal{U})$ contains the set

$$\mathbf{M}_0(\mathcal{U}) = \{ \delta_B \in \mathbf{M}(\mathcal{U}) : B \in \mathcal{U} \},\$$

of degenerate finitely additive probabilities, where δ_B is the finitely additive probability such that $\delta_B(\{B\}) = 1$. Let us notice that $\mathbf{M}(\mathcal{U})$ is closed under the following operations:

- **Countable convex combination:** For every $\mu_1, \mu_2, \ldots \in \mathbf{M}(\mathcal{U})$ and $\alpha_1, \alpha_2, \ldots \in [0, 1]$ with $\sum_{n=1}^{\infty} \alpha_n = 1$, the finitely additive probability $\sum_{n=1}^{\infty} \alpha_n \mu_n$ defined, for every $A \in \mathcal{P}(\mathcal{U})$, as $\sum_{n=1}^{\infty} \alpha_n \mu_n(A)$, belongs to $\mathbf{M}(\mathcal{U})$.
- **Conditioning:** For every $\mu \in \mathbf{M}(\mathcal{U})$ and $H \in \mathcal{P}(\mathcal{U})$ with $\mu(H) > 0$, the finitely additive probability μ_H defined, for every $A \in \mathcal{P}(\mathcal{U})$, as $\mu_H(A) = \frac{\mu(A \cap H)}{\mu(H)}$, belongs to $\mathbf{M}(\mathcal{U})$.

For every $H \in \mathcal{P}(S)^0$, the relation \preceq_H determines a relation \trianglelefteq_H on $\mathbf{M}(\mathcal{U})$ through constant acts defined, for every $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$, as

$$\mu_1 \leq_H \mu_2 \iff \overline{\mu_1} \precsim_H \overline{\mu_2}$$

In turn, the relation \leq_H determines a relation \leq_H^{\bullet} on \mathcal{U} defined, for every $A, B \in \mathcal{U}$, as

$$A \leq^{\bullet}_{H} B \Longleftrightarrow \delta_{A} \trianglelefteq_{H} \delta_{B}$$

Finally, the relation \leq_{H}^{\bullet} induces a relation \leq_{H}^{*} on X defined, for every $x, y \in X$, as

$$x \leq_H^* y \Longleftrightarrow \{x\} \leq_H^{\bullet} \{y\}.$$

For $H \in \mathcal{P}(S)^0$, if there are $x_*, x^* \in X$ such that $x_* <^*_H x^*$ then, for every $E \in \mathcal{P}(S)$, define the act

$$\mathbf{1}_{E}(s) = \begin{cases} \delta_{\{x^{*}\}} & \text{if } s \in E, \\ \delta_{\{x_{*}\}} & \text{if } s \notin E. \end{cases}$$
(2)

For every fixed $H \in \mathcal{P}(S)^0$, two acts $f, g \in \mathcal{F}$ are *comonotonic* if there are no $s, t \in S$ such that $f(s) \triangleleft_H f(t)$ and $g(t) \triangleleft_H g(s)$, where \triangleleft_H is the asymmetric part of the relation \trianglelefteq_H induced by \precsim_H on $\mathbf{M}(\mathcal{U})$ through constant acts.

Here, we assume that "subjective" uncertainty is modeled through a full conditional belief function according to Definition 1.

We look for a representation in terms of a conditional functional $\underline{\Lambda}$ defined, for every $f \in \mathcal{F}$ and every $H \in \mathcal{P}(S)^0$, as

$$\underline{\Lambda}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi(\mathrm{d}s|H),$$
(3)

where $\varphi : \mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0, 1]$ is a full conditional belief function according to Definition 1, $u : X \to \mathbb{R}$ is a bounded utility function and $\alpha : \mathcal{U} \to [0, 1]$ is a function that can be interpreted as a pessimism index function [17], which depends on the menu B. To simplify computations, denoting

$$V(f(s)) = \int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B), \quad (4)$$

we can write

$$\underline{\Lambda}(f|H) = \oint_{S} V(f(s))\varphi(\mathrm{d}s|H).$$
(5)

For every $H \in \mathcal{P}(S)^0$, the (conditional) functional $\underline{\Lambda}(\cdot|H)$ turns out to be a "subjective" conditional Choquet expectation of a state-contingent "objective" mixture of convex combinations of infima and suprema of utility on menus. We point out the resemblance of the internal menu-contingent convex combination with the Hurwicz's criterion [29].

More in detail, for every menu $B \in \mathcal{U}$, the corresponding utility is obtained as a convex combination of the "worst" and "best" utility values on the related consequences, where the weight of the convex combination is $\alpha(B)$. This gives rise to a utility function on menus that captures the attitude of the decision maker towards "objective" ambiguity, through the function α . Then, for every state $s \in S$, we take an "objective" expectation with respect to the finitely additive probability measure f(s) defined on menus that, in turn, gives rise to a state-contingent "objective" expected utility. Finally, we take a "subjective" conditional Choquet expectation with respect to the conditional belief function $\varphi(\cdot|H)$, defined on states. Notice that, due to Definition 1, $\Lambda(f|H)$ is well-defined even though the *ex ante* belief function $\varphi(H|S)$ is equal to 0. Further, the properties of the Choquet integral imply that the external conditional Choquet expectation is actually a lower conditional expectation computed with respect to $\operatorname{core}(\varphi(\cdot|H))$. In other terms, a decision maker deciding according to the conditional functional $\underline{\Lambda}(\cdot|\cdot)$ has a systematically pessimistic attitude towards "subjective" uncertainty.

The conditional functional $\underline{\Lambda}$ is said to *represent* the family of preference relations $\{ \preceq_H \}_{H \in \mathcal{P}(S)^0}$ if, for every $H \in \mathcal{P}(S)^0$ and every $f, g \in \mathcal{F}$, it holds that

$$f \preceq_H g \iff \underline{\Lambda}(f|H) \le \underline{\Lambda}(g|H).$$
 (6)

Our aim is to study the representability of $\{ \preceq_H \}_{H \in \mathcal{P}(S)^0}$ through a conditional functional $\underline{\Lambda}(\cdot|\cdot)$. For this, consider the following axioms.

(A1) Weak order: For every $H \in \mathcal{P}(S)^0$, \preceq_H is a weak order on \mathcal{F} .

(A2) Continuity: For every $H \in \mathcal{P}(S)^0$, for every $f, g, h \in \mathcal{F}$, if $f \prec_H g \prec_H h$, there exist $\alpha, \beta \in (0, 1)$ such that

$$\alpha f + (1 - \alpha)h \prec_H g \prec_H \beta f + (1 - \beta)h.$$

(A3) Comonotonic independence: For every $H \in \mathcal{P}(S)^0$, for every pairwise comonotonic $f, g, h \in \mathcal{F}$ and for every $\alpha \in (0, 1)$

$$f \preceq_H g \iff \alpha f + (1 - \alpha)h \preceq_H \alpha g + (1 - \alpha)h.$$

- (A4) Monotonicity: For every $H \in \mathcal{P}(S)^0$, for every $f, g \in \mathcal{F}$, if $f(s) \leq_H g(s)$, for all $s \in S$, then $f \preceq_H g$.
- (A5) Non-triviality: For every $H \in \mathcal{P}(S)^0$, there exist $f, g \in \mathcal{F}$ such that $f \prec_H g$.
- (A6) Complete monotonicity: For every $H \in \mathcal{P}(S)^0$, for every $k \ge 2$ and every $E_1, \ldots, E_k \in \mathcal{P}(S)$, if $x_*, x^* \in X$ are such that $x_* <_H x^*$ and for all $\emptyset \neq I \subseteq \{1, \ldots, k\}$

$$\mathbf{1}_{\bigcup_{i=1}^{k}E_{i}}\sim_{H}\overline{\mu_{\{1,\ldots,k\}}^{\cup}}\text{ and }\mathbf{1}_{\bigcap_{i\in I}E_{i}}\sim_{H}\overline{\mu_{I}^{\cap}}$$

with $\mu_{\{1,\ldots,k\}}^{\cup}, \mu_I^{\cap} \in \mathbf{M}(\mathcal{U})$, then the acts

$$g = \frac{1}{2^{k-1}} \left(\frac{\overline{\mu_{\{1,\dots,k\}}^{\cup}} + \sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\} \\ |I| \text{ even}}} \overline{\mu_I^{\cap}} \right) \text{ and } h = \frac{1}{2^{k-1}} \left(\sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\} \\ |I| \text{ odd}}} \overline{\mu_I^{\cap}} \right)$$

are such that

$$h \precsim_H g.$$

- (A7) Relevance: For every $H \in \mathcal{P}(S)^0$, for every $f, g \in \mathcal{F}$ with f(s) = g(s), for all $s \in H$, then $f \sim_H g$.
- (A8) Chain rule: For every $A \in \mathcal{P}(S)$ and $B, C \in \mathcal{P}(S)^0$ with $A \subseteq B \subseteq C$, for every $\mu_1, \mu_2, \mu_3 \in \mathbf{M}(\mathcal{U})$ with $\mu_1 \leq_B \mu_2 \leq_B \mu_3$ and $\mu_1 \leq_C \mu_2 \leq_C \mu_3$, if $f, g \in \mathcal{F}$ are such that $f_{|A} = \overline{\mu_3}_{|A}, f_{|A^c} = \overline{\mu_1}_{|A^c}, g_{|B} = \overline{\mu_2}_{|B}$ and $g_{|B^c} = \overline{\mu_1}_{|B^c}$, then

$$f \sim_B g \Longrightarrow f \sim_C g.$$

(A9) Scenario neutrality: For every $H, K \in \mathcal{P}(S)^0$ and for every $\overline{\mu_1}, \overline{\mu_2} \in \mathcal{F}_{\text{const}}$,

$$\overline{\mu_1} \precsim_H \overline{\mu_2} \Longleftrightarrow \overline{\mu_1} \precsim_K \overline{\mu_2}$$

(A10) Sure-thing: For every $H \in \mathcal{P}(S)^0$, for every $\mathcal{B} \in \mathcal{P}(\mathcal{U})$, for every $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$ it holds that

$$\begin{cases} \mu_1(\mathcal{B}) = 1, \overline{\mu_2} \prec_H \overline{\delta_B} \text{ for all } B \in \mathcal{B} \Longrightarrow \overline{\mu_2} \precsim_H \overline{\mu_1}; \\ \mu_1(\mathcal{B}) = 1, \overline{\delta_B} \prec_H \overline{\mu_2} \text{ for all } B \in \mathcal{B} \Longrightarrow \overline{\mu_1} \precsim_H \overline{\mu_2}. \end{cases}$$

(A11) Contingencywise dominance: For every $H \in \mathcal{P}(S)^0$, for every $B \in \mathcal{U}$, for every $\mu \in \mathbf{M}(\mathcal{U})$ whose corresponding belief function φ_{μ} on $\mathcal{P}(X)$ is a finitely additive probability, it holds that

$$\left\{ \begin{array}{l} \overline{\delta_{\{x\}}} \prec_H \overline{\mu} \text{ for all } x \in B \Longrightarrow \overline{\delta_B} \precsim_H \overline{\mu}; \\ \overline{\mu} \prec_H \overline{\delta_{\{x\}}} \text{ for all } x \in B \Longrightarrow \overline{\mu} \precsim_H \overline{\delta_B}. \end{array} \right.$$

(A12) Outcome boundedness: For every $H \in \mathcal{P}(S)^0$, for every $B \in \mathcal{U}$, there exist $x_1, x_2 \in X$ such that

$$\overline{\delta_{\{x_1\}}} \precsim_H \overline{\delta_B} \precsim_H \overline{\delta_{\{x_2\}}}.$$

Axioms (A1)–(A5) are the usual Schmeidler's adaptations of Anscombe-Aumann axioms [43] in the formulation of [20], stated for generalized Anscombe-Aumann acts and every preference relation in $\{ \preceq_H \}_{H \in \mathcal{P}(S)^0}$. Such axioms essentially deal with the external Choquet integral.

Axiom (A6) assures the complete monotonicity of the "subjective" conditional capacity used in the external Choquet integral. In particular, due to the properties of the Choquet integral [10, 25, 42], this consists in a conditional lower expectation computed with respect to $\operatorname{core}(\varphi(\cdot|H))$. We point out that requiring the property in (A6) to hold for every $2 \le k \le n$ with a fixed $n \ge 2$ we get a conditional *n*-monotone capacity. In particular, for n = 2 the corresponding axiom turns out to be equivalent, in presence of other axioms, to the *uncertainty aversion* axiom of Schmeidler [43], for every $H \in \mathcal{P}(S)^0$. We further notice that if (A6) is strengthened by requiring $f \sim_H g$, then $\varphi(\cdot|\cdot)$ reduces to a full conditional probability on $\mathcal{P}(S)$. Therefore, the strengthened version of (A6) plus axiom (A3), in presence of other axioms, imply the *independence axiom* of Schmeidler [43], for every $H \in \mathcal{P}(S)^0$. Axioms (A7)–(A8) cope with conditioning. In particular, axiom (A7) expresses a focusing conditioning rule, i.e., it states that in conditioning on H, only the part of acts inside of H counts. Such axiom is also referred to as *consequentialism* in the literature (see, e.g., [18]) but we maintain the name relevance as in [36].

Axiom (A8) copes with relating different conditioning events through a qualitative chain rule. In particular, axiom (A8) connects the *ex ante* preference relation \preceq_S with any other preference relation \preceq_H relating indifferences of \preceq_H on suitable acts to indifferences of \preceq_S . We point out that the implication in axiom (A8) cannot be reverted and is the key feature of the axiomatic notion of conditioning captured by Definition 1 that allows conditioning also on events H such that $\varphi(H|S) = 0$.

Example 2. Let $S = \{s_1, s_2, s_3\}$ and $X = \{x_1, x_2, x_3\}$, and take the full conditional probability $P(\cdot|\cdot)$ on $\mathcal{P}(S)$ such that

$\mathcal{P}(S)$	Ø	$\{s_1\}$	$\{s_2\}$	$\{s_3\}$	$\{s_1, s_2\}$	$\{s_1, s_3\}$	$\{s_2, s_3\}$	S
$P(\cdot S)$	0	0	0	1	0	1	1	1
$P(\cdot \{s_1, s_2\})$	0	0.5	0.5	0	1	0.5	0.5	1

while all other values can be recovered through ratios. We have that $P(\cdot|\cdot)$ is also a full conditional belief function on $\mathcal{P}(S)$ according to Definition 1. Let α be an arbitrary function $\alpha : \mathcal{U} \to [0,1]$ and u be such that $u(x_1) = 0$, $u(x_2) = 2$ and $u(x_3) = 3$.

Considering the acts

$$\begin{array}{c|cccc} S & s_1 & s_2 & s_3 \\ \hline f & \delta_{\{x_3\}} & \delta_{\{x_1\}} & \delta_{\{x_1\}} \\ g & \delta_{\{x_2\}} & \delta_{\{x_2\}} & \delta_{\{x_2\}} \end{array}$$

since, for i = 1, 2, 3, it holds that

$$\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) \delta_{\{x_i\}}(\mathrm{d}B) = u(x_i),$$

by equation (4) we have that

$$\begin{array}{c|cccc} S & s_1 & s_2 & s_3 \\ \hline V(f(s)) & 3 & 0 & 0 \\ V(g(s)) & 2 & 2 & 0 \\ \end{array}$$

Let $A = \{s_1\}$, $B = \{s_1, s_2\}$ and C = S. By equation (5) and the additivity of $P(\cdot|C)$ and $P(\cdot|B)$ we get

$$\begin{split} \underline{\Lambda}(f|C) &= \oint_{S} V(f(s))P(\mathrm{d}s|C) = \sum_{s \in S} V(f(s))P(\{s\}|C) = 0, \\ \underline{\Lambda}(g|C) &= \oint_{S} V(g(s))P(\mathrm{d}s|C) = \sum_{s \in S} V(g(s))P(\{s\}|C) = 0, \\ \underline{\Lambda}(f|B) &= \oint_{S} V(f(s))P(\mathrm{d}s|B) = \sum_{s \in S} V(f(s))P(\{s\}|B) = 1.5, \\ \underline{\Lambda}(g|B) &= \oint_{S} V(g(s))P(\mathrm{d}s|B) = \sum_{s \in S} V(g(s))P(\{s\}|B) = 2. \end{split}$$

Hence, it holds that

$$\underline{\Lambda}(f|C) = \underline{\Lambda}(g|C),$$

$$\underline{\Lambda}(f|B) < \underline{\Lambda}(g|B).$$

Referring to the family of preference relations $\{ \precsim_H \}_{H \in \mathcal{P}(S)^0}$ on $\mathcal{F} = \mathbf{M}(\mathcal{U})^S$ induced by $\underline{\Lambda}(\cdot|\cdot)$, we have that acts f and g satisfy the conditions in **(A8)** with A, B and C as above, but $f \sim_C g$ does not imply $f \sim_B g$.

Axiom (A9) assures the independence of the utility function on the conditioning event. Axiom (A10) is a restating of condition A4a in [15] assuring the boundedness of the utility and the inner integral expression. Finally, axioms (A11)–(A12) are inspired to Axiom 8 and Axiom 9 of [17], respectively, and guarantee the inner convex combination expression.

The following theorem shows that axioms (A1)–(A12) are sufficient to get a representation by a conditional functional $\underline{\Lambda}(\cdot|\cdot)$.

Theorem 1. If the family of relations $\{ \preceq_H \}_{H \in \mathcal{P}(S)^0}$ on \mathcal{F} satisfies $(\mathbf{A1}) - (\mathbf{A12})$ then there exist a full conditional belief function $\varphi : \mathcal{P}(S) \times \mathcal{P}(S)^0 \rightarrow [0,1]$, a non-constant bounded utility function $u : X \rightarrow \mathbb{R}$, and a pessimism index function $\alpha : \mathcal{U} \rightarrow [0,1]$ such that, for every $H \in \mathcal{P}(S)^0$, defining for every $f \in \mathcal{F}$,

$$\underline{\Lambda}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi(\mathrm{d}s|H),$$

for every $f, g \in \mathcal{F}$ it holds that

$$f\precsim_H g \Longleftrightarrow \underline{\Lambda}(f|H) \leq \underline{\Lambda}(g|H)$$

Moreover, φ is unique, u is unique up to positive linear transformations, and α is uniquely defined whenever $\inf_{x \in B} u(x) < \sup_{x \in B} u(x)$. The detailed proof of the above result is reported in Appendix A. In such proof, axioms (A1)–(A5) are used, for every $H \in \mathcal{P}(S)^0$, to get a nonconstant *affine* function $V_H : \mathbf{M}(\mathcal{U}) \to \mathbb{R}$ and a capacity $\varphi_H : \mathcal{P}(S) \to [0, 1]$ such that the functional defined, for all $f \in \mathcal{F}$, as

$$\underline{\Lambda}_{H}(f) = \oint_{S} V_{H}(f(s))\varphi_{H}(\mathrm{d}s),$$

represents the preference \preceq_H . Affinity of V_H means that

$$V_H(\alpha \mu_1 + (1 - \alpha)\mu_2) = \alpha V_H(\mu_1) + (1 - \alpha)V_H(\mu_2),$$

for every $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$ and $\alpha \in [0, 1]$. In particular, V_H is determined by the relation \preceq_H on constant acts, thus it does not depend on "subjective" uncertainty.

In turn, axiom (A10) is used to get an integral expression of V_H through a non-constant bounded utility function $v_H : \mathcal{U} \to \mathbb{R}$ on menus. Then, axioms (A11) and (A12) are used to build a non-constant bounded utility function $u_H : X \to \mathbb{R}$ on consequences such that v_H satisfies internality with respect to u_H , i.e., for every $B \in \mathcal{U}$,

$$\inf_{x \in B} u_H(x) \le v_H(B) \le \sup_{x \in B} u_H(x).$$

Note that $\alpha_H(B) \in [0, 1]$ is a pessimism index allowing to write $v_H(B)$ as the convex combination of the "worst" and "best" utility values in the menu B, that is

$$v_H(B) = \alpha_H(B) \cdot \inf_{x \in B} u_H(x) + (1 - \alpha_H(B)) \cdot \sup_{x \in B} u_H(x).$$

Now, axiom (A9) is used to get the independence of V_H , v_H , u_H and α_H of the conditioning event $H \in \mathcal{P}(S)^0$. Thus, for all conditioning events we can take the same functions V, v, u and α , respectively.

Finally, axioms (A6)–(A8) are used to show that the function φ : $\mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0,1]$ obtained setting, for every $E|H \in \mathcal{P}(S) \times \mathcal{P}(S)^0$,

$$\varphi(E|H) = \varphi_H(E),$$

is a full conditional belief function according to Definition 1.

The sketched construction shows that u and α (and, so, v and V) only depend on "objective" uncertainty, since they are determined by the relations on constant acts. In particular, u and α do not depend on "subjective" uncertainty which is encoded in the full conditional belief function $\varphi(\cdot|\cdot)$

on $\mathcal{P}(S)$. Hence, the three parameters of the model u, α and φ clearly separate the treatment of "objective" and "subjective" ambiguity. More precisely, "objective" ambiguity is dealt with a precise utility function u on consequences in X endowed with a pessimism index function α . By contrast, (conditional) "subjective" ambiguity is dealt with the belief function $\varphi(\cdot|H)$ on $\mathcal{P}(S)$, i.e., with the set of finitely additive probability measure **core**($\varphi(\cdot|H)$) on $\mathcal{P}(S)$.

The following proposition states that all axioms (A1)–(A11) are necessary to get a representation of $\{ \preceq_H \}_{H \in \mathcal{P}(S)^0}$, through a conditional functional $\underline{\Lambda}(\cdot | \cdot)$ of the form discussed above.

Proposition 1. If there exist a full conditional belief function $\varphi : \mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0,1]$, a non-constant bounded utility function $u : X \to \mathbb{R}$, and a pessimism index function $\alpha : \mathcal{U} \to [0,1]$ such that the conditional functional $\underline{\Lambda}(\cdot|\cdot)$ defined as in (3) represents the family of relations $\{\preceq_H\}_{H\in\mathcal{P}(S)^0}$, then the family $\{\preceq_H\}_{H\in\mathcal{P}(S)^0}$ satisfies (A1)–(A11).

We stress that, axiom (A12) is not necessary as shown by the following example. On the other hand, axiom (A12) is vacuously satisfied if X is finite, thus in that case all axioms are necessary and sufficient. Thus, it follows that (A12) has a "regularity" role when the set of state is not finite.

Example 3. Take $S = X = \mathbb{N} = \{1, 2, ...\}$, $u(x) = 1 - \frac{1}{x}$, α constantly equal to 0 on \mathcal{U} , and φ any full conditional belief function on $\mathcal{P}(S)$, we have that

$$\underline{\Lambda}(\overline{\delta_X}|S) = \sup_{x \in X} u(x) = 1,$$

but there is no $x_2 \in X$ such that $\underline{\Lambda}(\overline{\delta_{\{x_2\}}}|S) = u(x_2) \geq 1$. Thus, we have that there is no $x_2 \in X$ such that $\overline{\delta_X} \preceq_S \overline{\delta_{\{x_2\}}}$ and axiom (A12) is not satisfied.

The conditional functional $\underline{\Lambda}(\cdot|\cdot)$ can be specialized by incorporating a systematically pessimistic or optimistic attitude towards "objective" ambiguity. This consists in selecting an α which is constantly equal to 1 or 0, respectively, and this is possible adding a further axiom.

In case of a systematically pessimistic attitude we search for a conditional

functional $\underline{\Phi}$ defined, for every $f \in \mathcal{F}$ and every $H \in \mathcal{P}(S)^0$, as

$$\underline{\Phi}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\inf_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi(\mathrm{d}s|H) \\ = \oint_{S} \left[\oint_{X} u(x)\varphi_{f(s)}(\mathrm{d}x) \right] \varphi(\mathrm{d}s|H) \\ = \oint_{S} \left[\min_{\nu \in \mathbf{core}(\varphi_{f(s)})} \int_{X} u(x)\nu(\mathrm{d}x) \right] \varphi(\mathrm{d}s|H), \quad (7)$$

where $\varphi_{f(s)}$ is the belief function on $\mathcal{P}(X)$ induced by the finitely additive probability f(s) on $\mathcal{P}(\mathcal{U})$. Thus, the internal integral reduces to a state-contingent "objective" lower expected utility. The following axiom is inspired to **Axiom 12** in [17].

(A13) Ignorance pessimism: For every $H \in \mathcal{P}(S)^0$, for every $A, B \in \mathcal{U}$, if for all $x \in A$, there exists $y \in B$ such that $\overline{\delta_{\{y\}}} \preceq_H \overline{\delta_{\{x\}}}$ then $\overline{\delta_B} \preceq_H \overline{\delta_A}$.

The above axiom requires that, for every pair of menus of consequences A, B and scenario H, if for every consequence x in A we can find a consequence y in B such that x selected with certainty is weakly preferred to y selected with certainty, then menu A selected with certainty must be weakly preferred to menu B selected with certainty.

Theorem 2. If the family of relations $\{ \precsim_H \}_{H \in \mathcal{P}(S)^0}$ on \mathcal{F} satisfies (A1)– (A12) and (A13) then there exist a full conditional belief function φ : $\mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0,1]$ and a non-constant bounded utility function $u : X \to \mathbb{R}$ such that, for every $H \in \mathcal{P}(S)^0$, defining for every $f \in \mathcal{F}$,

$$\underline{\Phi}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\inf_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi(\mathrm{d}s|H),$$

for every $f, g \in \mathcal{F}$ it holds that

$$f \precsim_H g \Longleftrightarrow \underline{\Phi}(f|H) \le \underline{\Phi}(g|H).$$

Moreover, φ is unique and u is unique up to positive linear transformations.

In case of a systematically optimistic attitude we search for a conditional

functional $\underline{\Psi}$ defined, for every $f \in \mathcal{F}$ and every $H \in \mathcal{P}(S)^0$, as

$$\underline{\Psi}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi(\mathrm{d}s|H) \\
= \oint_{S} \left[\oint_{X} u(x) \psi_{f(s)}(\mathrm{d}x) \right] \varphi(\mathrm{d}s|H) \\
= \oint_{S} \left[\max_{\nu \in \mathbf{core}(\varphi_{f(s)})} \int_{X} u(x) \nu(\mathrm{d}x) \right] \varphi(\mathrm{d}s|H), \quad (8)$$

where $\psi_{f(s)}$ is the dual of the belief function $\varphi_{f(s)}$. Thus, the internal integral reduces to a state-contingent "objective" upper expected utility. The following axiom is inspired to **Axiom 12** in [17].

(A13') Ignorance optimism: For every $H \in \mathcal{P}(S)^0$, for every $A, B \in \mathcal{U}$, if for all $y \in B$, there exists $x \in A$ such that $\overline{\delta_{\{y\}}} \preceq_H \overline{\delta_{\{x\}}}$ then $\overline{\delta_B} \preceq_H \overline{\delta_A}$.

The above axiom requires that, for every pair of menus of consequences A, B and scenario H, if for every consequence y in B we can find a consequence x in A such that x selected with certainty is weakly preferred to y selected with certainty, then menu A selected with certainty must be weakly preferred to menu B selected with certainty.

Theorem 3. If the family of relations $\{\preceq_H\}_{H \in \mathcal{P}(S)^0}$ on \mathcal{F} satisfies (A1)–(A12) and (A13') then there exist a full conditional belief function φ : $\mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0,1]$ and a non-constant bounded utility function $u: X \to \mathbb{R}$ such that, for every $H \in \mathcal{P}(S)^0$, defining for every $f \in \mathcal{F}$,

$$\underline{\Psi}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi(\mathrm{d}s|H),$$

for every $f, g \in \mathcal{F}$ it holds that

$$f \precsim_H g \Longleftrightarrow \underline{\Psi}(f|H) \le \underline{\Psi}(g|H).$$

Moreover, φ is unique and u is unique up to positive linear transformations.

The following proposition states that axioms (A1)–(A11) and (A13) [(A13')] are necessary to get a representation of $\{ \preceq_H \}_{H \in \mathcal{P}(S^0)}$ through a conditional functional $\underline{\Phi}(\cdot|\cdot)$ [$\underline{\Psi}(\cdot|\cdot)$]. Also in this case, axiom (A12) is only sufficient, in general, but becomes necessary when X is finite.

Proposition 2. If there exist a full conditional belief function $\varphi : \mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0,1]$ and a non-constant bounded utility function $u: X \to \mathbb{R}$ such that the conditional functional $\underline{\Phi}(\cdot|\cdot) / \underline{\Psi}(\cdot|\cdot)]$ defined as in (7) [(8)] represents the family of relations $\{ \precsim_H \}_{H \in \mathcal{P}(S)^0}$, then the family $\{ \precsim_H \}_{H \in \mathcal{P}(S)^0}$ satisfies (A1)-(A11) and (A13) [(A13')].

We point out that a family of preference relations cannot satisfy simultaneously (A1)–(A12), (A13) and (A13'). Indeed, this would lead to a constant utility function $u: X \to \mathbb{R}$ but this is in contrast with axiom (A5).

4.1. A systematically optimistic attitude towards subjective uncertainty

The three conditional functionals $\underline{\Lambda}(\cdot|\cdot)$, $\underline{\Phi}(\cdot|\cdot)$ and $\underline{\Psi}(\cdot|\cdot)$ introduced so far assume a systematically pessimistic attitude in "subjective" uncertainty evaluations. A completely opposite behavior is encoded in the conditional functionals defined, for every $f \in \mathcal{F}$ and $H \in \mathcal{P}(S)^0$, as

$$\overline{\Lambda}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \psi(\mathrm{d}s|H),$$
(9)

$$\overline{\Phi}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\inf_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \psi(\mathrm{d}s|H), \tag{10}$$

$$\overline{\Psi}(f|H) = \oint_{S} \left[\int_{\mathcal{U}} \left(\sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \psi(\mathrm{d}s|H), \tag{11}$$

where $\psi : \mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0, 1]$ is a full conditional plausibility function according to Definition 2.

We focus on the conditional functional $\overline{\Lambda}(\cdot|\cdot)$, of which $\overline{\Phi}(\cdot|\cdot)$ and $\overline{\Psi}(\cdot|\cdot)$ are particular cases. Also in this case, the internal integral gives rise to a utility function on menus that captures the attitude of the decision maker towards "objective" ambiguity, relying on the function α . Then, we take a "subjective" conditional Choquet expectation with respect to the conditional plausibility function $\psi(\cdot|H)$, defined on states. Again, due to Definition 2, $\overline{\Lambda}(f|H)$ is well-defined even though the *ex ante* plausibility function $\psi(H|S)$ is equal to 0. Further, the external conditional Choquet expectation is actually an upper conditional expectation computed with respect to the core of the dual belief function of $\psi(\cdot|H)$. In other terms, a decision maker deciding according to the conditional functional $\overline{\Lambda}(\cdot|\cdot)$ has a systematically optimistic attitude towards "subjective" uncertainty.

Further, recalling (4), we can simplify computations by writing

$$\overline{\Lambda}(f|H) = \oint_{S} V(f(s))\psi(\mathrm{d}s|H).$$
(12)

To get representation theorems for conditional functionals $\overline{\Lambda}(\cdot|\cdot)$, $\overline{\Phi}(\cdot|\cdot)$ and $\overline{\Psi}(\cdot|\cdot)$ analogous to Theorems 1, 2 and 3 it is sufficient to replace axiom (A6) with

(A6') Complete alternance: For every $H \in \mathcal{P}(S)^0$, for every $k \ge 2$ and every $E_1, \ldots, E_k \in \mathcal{P}(S)$, if $x_*, x^* \in X$ are such that $x_* <_H x^*$ and for all $\emptyset \ne I \subseteq \{1, \ldots, k\}$

$$\mathbf{1}_{\bigcap_{i=1}^{k} E_{i}} \sim_{H} \overline{\mu_{\{1,\dots,k\}}^{\cap}} \text{ and } \mathbf{1}_{\bigcup_{i \in I} E_{i}} \sim_{H} \overline{\mu_{I}^{\cup}}$$

with $\mu_{\{1,\ldots,k\}}^{\cap}, \mu_I^{\cup} \in \mathbf{M}(\mathcal{U})$, then the acts

$$g = \frac{1}{2^{k-1}} \left(\frac{\overline{\mu_{\{1,\dots,k\}}^{\cap}} + \sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\}\\|I| \text{ even}}} \overline{\mu_I^{\cup}} \right) \text{ and } h = \frac{1}{2^{k-1}} \left(\sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\}\\|I| \text{ odd}}} \overline{\mu_I^{\cup}} \right)$$

are such that

$$g \precsim_H h$$

Also for axiom (A6') we have that requiring the property to hold for every $2 \leq k \leq n$ with a fixed $n \geq 2$ we get a conditional *n*-alternating capacity. In particular, for n = 2 the corresponding axiom turns out to be equivalent, in presence of other axioms, to the *uncertainty appeal* axiom of Schmeidler [43], for every $H \in \mathcal{P}(S)^0$.

4.2. A Bayesian decision maker

So far we have assumed that "subjective" uncertainty is not necessarily additive. Now, we aim to model a Bayesian decision maker whose "subjective" uncertainty is unambiguous and can be expressed by a full conditional probability. We then consider the following strengthening of axiom (A3)

(A3') Independence: For every $H \in \mathcal{P}(S)^0$, for every $f, g, h \in \mathcal{F}$ and for every $\alpha \in (0, 1)$

$$f \preceq_H g \iff \alpha f + (1 - \alpha)h \preceq_H \alpha g + (1 - \alpha)h.$$

If in the previous representation theorems (see [43]) we substitute axiom (A3) with (A3'), then $\varphi(\cdot|\cdot) = \psi(\cdot|\cdot) = P(\cdot|\cdot)$ where $P(\cdot|\cdot)$ is a full conditional probability on $\mathcal{P}(S)$, that is we have a decision maker with unambiguous "subjective" beliefs. We stress that, in presence of other axioms, axiom (A3') alone implies the three axioms (A3), (A6) and (A6'). Further, still in presence of other axioms, (A3') also implies the following (see [47] for an analysis of dynamic consistency)

(A14) Strong dynamic consistency: For every $f, g \in \mathcal{F}$ and every $H, K \in \mathcal{P}(S)^0$, if $f \preceq_H g, f \preceq_K g$, and $H \cap K = \emptyset$ then $f \preceq_{H \cup K} g$, where the last preference is strict if both the first two are.

In turn, property (A14) implies the weaker

(A14') Weak dynamic consistency: For every $f, g \in \mathcal{F}$ and every $H, K \in \mathcal{P}(S)^0$, if $f \preceq_H g, f \preceq_K g$, and $H \cap K = \emptyset$ then $f \preceq_{H \cup K} g$.

We stress that axiom (A14') is generally not satisfied (and so neither is (A14)) by conditional preferences representable by a conditional functional $\underline{\Lambda}(\cdot|\cdot)$ or $\overline{\Lambda}(\cdot|\cdot)$ as the following example shows. In other terms, in presence of other axioms, the combination of (A3) and (A6) or (A3) and (A6') does not imply (A14') (and so neither (A14)).

Example 4. Let $S = \{s_1, s_2, s_3\}$ and $X = \{x_1, x_2, x_3\}$ and consider the full conditional belief function on $\mathcal{P}(S)$ such that

$$\begin{array}{c|c|c|c|c|c|c|c|c|} \mathcal{P}(S) & \emptyset & \{s_1\} & \{s_2\} & \{s_3\} & \{s_1,s_2\} & \{s_1,s_3\} & \{s_2,s_3\} & S \\ \hline \varphi(\cdot|S) & 0 & 0.2 & 0.2 & 0.1 & 0.4 & 0.6 & 0.3 & 1 \\ \end{array}$$

and, for every $E|H \in \mathcal{P}(S) \times \mathcal{P}(S)^0$, $\varphi(E|H) = \frac{\varphi(E \cap H|S)}{\varphi(H|S)}$.

Let α be an arbitrary function $\alpha : \mathcal{U} \to [0, 1]$ and u be such that $u(x_1) = 0$, $u(x_2) = 1$, $u(x_3) = 2$. Consider the acts f and g defined as

S	s_1	s_2	s_3
f	$\delta_{\{x_2\}}$	$\delta_{\{x_2\}}$	$\delta_{\{x_2\}}$
g	$\delta_{\{x_3\}}$	$\delta_{\{x_3\}}$	$\delta_{\{x_1\}}$

Since, for i = 1, 2, 3, it holds that

$$\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) \delta_{\{x_i\}}(\mathrm{d}B) = u(x_i),$$

for $H = \{s_2, s_3\}$ and $H^c = \{s_1\}$, simple computations show that

$$\begin{split} \underline{\Lambda}(f|H) &= 1 \quad < \quad 1.\overline{3} = \underline{\Lambda}(g|H), \\ \underline{\Lambda}(f|H^c) &= 1 \quad < \quad 2 = \underline{\Lambda}(g|H^c), \\ \underline{\Lambda}(f|S) &= 1 \quad > \quad 0.8 = \underline{\Lambda}(g|S), \end{split}$$

hence the family of preference relations induced by $\underline{\Lambda}(\cdot|\cdot)$ does not satisfy (A14').

Now, for every $E \in \mathcal{P}(S)$, define $\psi(E|S) = 1 - \varphi(E^c|S)$ that is

and, for every $E|H \in \mathcal{P}(S) \times \mathcal{P}(S)^0$, $\psi(E|H) = \frac{\psi(E \cap H|S)}{\psi(H|S)}$. The function $\psi(\cdot|\cdot)$ is a conditional plausibility function according to Definition 2. Take α , u as before, and let $\mu = 0.8 \cdot \delta_{\{x_2\}} + 0.2 \cdot \delta_{\{x_1\}}$. Consider the act h defined as

Since it holds that

$$\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) \mu(\mathrm{d}B) = 0.8$$

referring to $\overline{\Lambda}(\cdot|\cdot)$ and taking $K = \{s_1, s_2\}$ and $K^c = \{s_3\}$, we have that

$$\begin{split} \overline{\Lambda}(h|K) &= 0.\overline{8} < 1 = \overline{\Lambda}(f|K), \\ \overline{\Lambda}(h|K^c) &= 0.8 < 1 = \overline{\Lambda}(f|K^c), \\ \overline{\Lambda}(h|S) &= 1.12 > 1 = \overline{\Lambda}(f|S), \end{split}$$

hence the family of preference relations induced by $\overline{\Lambda}(\cdot|\cdot)$ does not satisfy (A14').

This example further shows that the violation of (A14') both for $\underline{\Lambda}(\cdot|\cdot)$ and $\overline{\Lambda}(\cdot|\cdot)$ is not necessarily due to events with null unconditional measure.

We point out that, even though axiom (A14') (and so (A14)) may fail to hold for the entire family of relations $\{ \precsim_{H} \}_{H \in \mathcal{P}(S)^{0}}$, it can happen that (A14') (or even (A14)) holds for all the relations in a subfamily $\{ \precsim_{H} \}_{H \in \mathcal{H}}$, where $\mathcal{H} \subseteq \mathcal{P}(S)^{0}$ is a non-empty set closed with respect to finite unions. For instance, this is the case if $\varphi(\cdot|H)$ is a finitely additive probability and $\mathcal{H} = \{ K \in \mathcal{P}(S)^{0} : K \subseteq H, \varphi(K|H) > 0 \}$. In particular, if $\varphi(\cdot|\cdot)$ is a full conditional probability on $\mathcal{P}(S)$ then the entire family of relations $\{ \precsim_{H \in \mathcal{P}(S)^{0} \text{ satisfies (A14)}.$

It is important to notice that, contrary to [47], axiom (A14) asks that both preferences $f \prec_H g$ and $f \prec_K g$ are strict for implying that $f \prec_{H\cup K} g$ is strict. Indeed, even restricting to conditional probability for expressing "subjective" uncertainty, it could happen that $f \sim_H g$, $f \prec_K g$ and $f \sim_{H\cup K} g$, as the following example shows. This is essentially due to the fact that we allow situations in which $P(H|H\cup K) = 0$ or $P(K|H\cup K) = 0$. **Example 5.** Let $S = \{s_1, s_2, s_3, s_4\}$ and $X = \{x_1, x_2, x_3\}$ and consider the full conditional probability $P(\cdot|\cdot)$ on $\mathcal{P}(S)$, where $H = \{s_1, s_2\}$ and $H^c = \{s_3, s_4\}$, such that

S	s_1	s_2	s_3	s_4
$P(\cdot S)$	0.5	0.5	0	0
$P(\cdot H^c)$	0	0	0.5	0.5

while all other values P(A|B) can be determined through ratios taking, respectively, $P(\cdot|S)$ if P(B|S) > 0 and $P(\cdot|H^c)$ otherwise.

Let α be an arbitrary function $\alpha : \mathcal{U} \to [0, 1]$ and u be such that $u(x_1) = 0$, $u(x_2) = 1$, $u(x_3) = 2$. Consider the acts f and g defined as

Since, for i = 1, 2, 3, it holds that

$$\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) \delta_{\{x_i\}}(\mathrm{d}B) = u(x_i),$$

simple computations show that

$$\underline{\Lambda}(f|H) = 1 = 1 = \underline{\Lambda}(g|H),$$

$$\underline{\Lambda}(f|H^c) = 0.5 < 1 = \underline{\Lambda}(g|H^c),$$

$$\underline{\Lambda}(f|S) = 1 = 1 = \underline{\Lambda}(g|S),$$

hence the family of preference relations induced by $\underline{\Lambda}(\cdot|\cdot)$ is such that $f \sim_H g$, $f \prec_{H^c} g$ and $f \sim_S g$.

5. A motivating decision problem (continuation)

We get back to the decision problem introduced in Section 2. We have that, for i = 1, 2, 3, the class of probabilities \mathbf{P}^i on $\mathcal{P}(X)$ gives rise to a probability measure μ_i on $\mathcal{P}(\mathcal{U})$, where $\mathcal{U} = \mathcal{P}(X) \setminus \{\emptyset\}$. In particular, the distribution m_i of μ_i on \mathcal{U} coincides with the Möbius inverse of a belief function φ_i on $\mathcal{P}(X)$ (see, e.g., [25]). Denoting $x_1 = -1000, x_2 = 0, x_3 =$ $1000, A_i = \{x_i\}$ and $A_{ij} = \{x_i, x_j\}$, we have

\mathcal{U}	A_1	A_2	A_3	A_{12}	A_{13}	A_{23}	X
m_1	0	0.5	0	0	0.5	0	0
m_2	0	0	0.3	0.7	0	0	0
m_3	0.3	0	0	0	0	0.7	0

Notice that each μ_i gives also rise to a plausibility function ψ_i on $\mathcal{P}(X)$ that is dual to φ_i .

Hence, identifying instrument i with the probability measure μ_i on $\mathcal{P}(\mathcal{U})$, the investment strategies f, g, h, l introduced in Section 2 reduce to the generalized Anscombe-Aumann acts

S	s_1	s_2	s_3	s_4
f	μ_2	μ_2	μ_2	μ_2
g	μ_1	μ_3	μ_3	μ_3
h	μ_3	μ_3	μ_2	μ_2
l	μ_3	μ_3	μ_3	μ_3

where, in particular, $f = \overline{\mu_2}$ and $l = \overline{\mu_3}$.

We stress that, assuming that the agent is a profit maximizer consists in selecting a strictly increasing utility function $u : X \to [0, 1]$, that is $u(x_1) < u(x_2) < u(x_3)$. Further, assuming that the agent is systematically pessimistic in coping with "objective" ambiguity consists in taking the pessimism index function $\alpha : \mathcal{U} \to [0, 1]$ constantly equal to 1.

For $C = \{s_1, s_2\}$ and $C^c = \{s_3, s_4\}$, given the preference pattern introduced in Section 2

$$f \sim_S g, \quad f \sim_S h, \quad f \prec_S l, \quad f \prec_C g, \quad \text{and} \quad f \prec_{C^c} g,$$

our aim is to show its representability with a conditional functional $\underline{\Lambda}(\cdot|\cdot)$, that, due to the assumption on α , reduces to $\underline{\Phi}(\cdot|\cdot)$.

Notice that statements $f \prec_S l$ and $f \sim_S h$ identify the event C as a "null" event for the agent. Indeed, if $\varphi(\cdot|\cdot)$ is a full conditional belief function on $\mathcal{P}(S)$ according to Definition 1, we have

$$\underline{\Phi}(f|S) = \underline{\Phi}(\overline{\mu_2}|S) = v_2 \quad \langle \quad v_3 = \underline{\Phi}(\overline{\mu_3}|S) = \underline{\Phi}(l|S),$$

$$\underline{\Phi}(f|S) = \underline{\Phi}(\overline{\mu_2}|S) = v_2 \quad = \quad (v_3 - v_2) \cdot \varphi(C|S) + v_2 = \underline{\Phi}(h|S),$$

which implies that $\varphi(C|S) = 0$. Moreover, the statements $f \sim_S g$, $f \prec_C g$, $f \prec_C g$, $f \prec_C g$ violate axiom (A14), thus $\varphi(\cdot|\cdot)$ cannot be a full conditional probability on $\mathcal{P}(S)$.

Denote $B_i = \{s_i\}, B_{ij} = \{s_i, s_j\}, B_{ijk} = \{s_i, s_j, s_k\}$. The above preferences can be represented taking $u(x_1) = 0, u(x_2) = 1, u(x_3) = 2$ together with the full conditional belief function $\varphi(\cdot|\cdot)$ on $\mathcal{P}(S)$ such that

$\mathcal{P}(S)$	Ø	B_1	B_2	B_3	B_4	B_{12}	B_{13}	B_{14}	B_{23}	B_{24}	B_{34}	B_{123}	B_{124}	B_{134}	B_{234}	S
$\varphi(\cdot S)$	0	0	0	0.25	0.25	0	0.25	0.25	0.25	0.25	0.5	0.25	0.25	0.5	0.5	1
$\varphi(\cdot C)$	0	0.25	0.75	0	0	1	0.25	0.25	0.75	0.75	0	1	1	0.25	0.75	1
$\varphi(\cdot C^c)$	0	0	0	0.5	0.5	0	0.5	0.5	0.5	0.5	1	0.5	0.5	1	1	1

while for every $E|H \in \mathcal{P}(S) \times \mathcal{P}(S)^0$ with $H \neq C$ and $H \neq C^c$, we set

$$\varphi(E|H) = \begin{cases} \frac{\varphi(E \cap H|C)}{\varphi(H|C)}, & \text{if } H \subseteq C, \\ \frac{\varphi(E \cap H|S)}{\varphi(H|S)}, & \text{otherwise.} \end{cases}$$

Notice that both $\varphi(\cdot|C)$ and $\varphi(\cdot|C^c)$ are probability measures on $\mathcal{P}(S)$, while $\varphi(\cdot|S)$ is not. The choice of the *u* above implies that

$$\int_{\mathcal{U}} \left(\inf_{x \in B} u(x) \right) \mu_i(\mathrm{d}B) = \begin{cases} 0.5 & \text{if } i = 1, \\ 0.6 & \text{if } i = 2, \\ 0.7 & \text{if } i = 3. \end{cases}$$

so, by (4), we get

S	s_1	s_2	s_3	s_4
V(f(s))	0.6	0.6	0.6	0.6
V(g(s))	0.5	0.7	0.7	0.7
V(h(s))	0.7	0.7	0.6	0.6
V(l(s))	0.7	0.7	0.7	0.7

Finally, simple computations show that

$$\begin{split} \underline{\Phi}(l|S) &= \underline{\Phi}(g|C^c) &= 0.7, \\ \underline{\Phi}(g|C) &= 0.65, \\ \underline{\Phi}(f|S) &= \underline{\Phi}(g|S) = \underline{\Phi}(h|S) = \underline{\Phi}(f|C) = \underline{\Phi}(f|C^c) &= 0.6, \end{split}$$

thus all the stated preferences are represented.

6. Conclusions

In this paper we assume that a decision maker is able to provide a family of conditional preference relations on acts mapping states of the world to finitely additive probabilities on menus of consequences. In particular, the latter can be seen as generalized lotteries conveying a form of "objective" ambiguity that reduces to belief/plausibility functions.

We first assume "subjective" uncertainty is ambiguous and modeled by a full conditional belief function. In this case, we propose a system of axioms that assure the representability of preferences by a "subjective" conditional Choquet expectation of a state-contingent "objective" mixture of convex combinations of infima and suprema of utility on menus. Then, we axiomatise a systematically pessimistic/optimistic attitude towards "objective" ambiguity. The last two models generalize to arbitrary sets of consequences and states of the world (besides incorporating "subjective" ambiguity), those introduced in [40] that, in turn, generalize the conditional version of the Anscombe-Aumann model given in [36] by introducing "objective" ambiguity.

This leads to consider a systematically optimistic approach to "subjective" ambiguity by referring to a conditional plausibility function. We stress that all the models we introduce allow conditioning on "null" (i.e., "unexpected") events, like a pandemic, characterized by a "null" *ex ante* measure.

Here we consider comparisons of acts only conditionally on the same event H, i.e., $f \preceq_H g$: a possible extension of our models could be to consider comparisons under different conditioning events $(f, H) \preceq (g, K)$ as done by Fishburn in [16] without taking care of ambiguity.

The main similarity of our models is with the *Choquet expected utility* (*CEU*) model by Schmeidler [42] and the maximin expected utility (*MEU*) model by Gilboa and Schmeidler [22]. Nevertheless, our models differ from them since they do not cope with conditioning and with "objective" ambiguity in the form addressed in the present paper. In particular, the form of "objective" ambiguity dealt with in the paper is of the same type of generalized lotteries due to Jaffray [30]. Still referring to the classical finite Anscombe-Aumann formulation without conditioning and "objective" ambiguity, in [19] Giang proposes a "subjective" foundation of possibility measures, that is maxitive plausibility functions.

In [48], limiting to a finite S and X, Vierø considers acts mapping states of the world to non-empty compact convex polyhedral sets of probability measures on X. If we restrict to the finite setting as well and, for every act f, we refer to **core**(f(s)) for $s \in S$, our decision objects turn out to be similar to those in [48]. Nevertheless, in [48] the author takes into account unconditional preferences only and refers to a representation functional in which "subjective" uncertainty is modeled with a probability measure. Further, she considers a state-contigent convex combination of expected utilities computed with respect to the "best" and "worst" lotteries in each set of probabilities given by an act on a state. Hence, the numerical reference model in [48] is quite different from ours.

In the paper [26] Gul and Pesendorfer consider Savage acts mapping the states of the world to a closed interval X of real numbers, expressing monetary payoffs. Limiting to an unconditional decision problem, the authors provide a characterization of a preference relation \preceq on the set of all such acts. Such characterization refers to a numeric representation of \preceq , namely *expected uncertainty utility (EUU)*, parametrized by an interval utility function u defined on the set of all closed sub-intervals of X and a countably additive complete and non-atomic prior probability μ on a suitable σ -algebra on the states. In detail, the proposed representation maps each act f to the expectation with respect to μ of u evaluated on the envelope of f. The main result in [26] requires an infinite state space. For this, the authors restrict to a finite state space S (seen as a partition of the original state space) and consider discrete acts defined on S. They show that the original prior μ induces a probability π on $\mathcal{P}(S)^0$. In this finite case, they get a representation that consists in an expectation with respect to π of the interval utility u evaluated on the infimum and supremum of f, varying $A \in \mathcal{P}(S)^0$. The probability π turns out to be the Möbius inverse of a belief function on $\mathcal{P}(S)$, thus the model in [26] provides a subjective foundation for the Dempster-Shafer theory of evidence. Our models work in a different setting with respect to [26] and consider conditional belief/plausibility functions according to Definitions 1 and 2 as primitive concepts for modeling "subjective" uncertainty.

The issue of modeling updating for ambiguous beliefs in terms of preferences has been considered by Pacheco Pires in [38], where the *full Bayesian conditioning rule* is taken as reference. Still referring to a generalized Bayesian conditioning rule, Ghirardato, Maccheroni and Marinacci in [18] deal with the issue of dynamic consistency advocating that such condition should hold only on acts that are not subject to "subjective" ambiguity.

We point out that, as already discussed in Section 4, the models proposed in this paper can be generalized by weakening axiom (A6) [(A6')] so as to obtain a "subjective" full conditional uncertainty measure $\varphi(\cdot|\cdot)$ on $\mathcal{P}(S)$ that is only 2-monotone [2-alternating]. Further, replacing axioms (A3) and (A6) with a straightforward adaptation of *C*-independence and uncertainty aversion [appeal] axioms of [22], the "subjective" full conditional uncertainty measure $\varphi(\cdot|\cdot)$ on $\mathcal{P}(S)$ is substituted by a family $\{\mathcal{C}_H\}_{H\in\mathcal{P}(S)^0}$ of non-empty closed convex sets of finitely additive probability measures on $\mathcal{P}(S)$. The family $\{\mathcal{C}_H\}_{H\in\mathcal{P}(S)^0}$ is used to compute a lower [upper] expectation, locally on every $H \in \mathcal{P}(S)^0$. Moreover, in presence of other axioms, $\{\mathcal{C}_H\}_{H\in\mathcal{P}(S)^0}$ is such that the corresponding family of lower [upper] envelopes obeys to a form of focusing and chain rule.

An interesting line for future research is the modeling of "objective" generalized lotteries in the framework of 2-monotone [2-alternating] capacities. Referring to results in [24, 35], in this case we should substitute the set $\mathbf{M}(\mathcal{U})$ with a (suitable) set of signed finitely additive measures.

Appendix A. Proofs of results

Proof of Theorem 1. For every $H \in \mathcal{P}(S)^0$, axioms (A1)–(A5) imply the existence of a non-constant affine function $V_H : \mathbf{M}(\mathcal{U}) \to \mathbb{R}$ and a capacity $\varphi_H : \mathcal{P}(S) \to [0, 1]$ such that, for every $f, g \in \mathcal{F}_{simple}$, it holds that

$$f \preceq_H g \iff \oint_S V_H(f(s))\varphi_H(\mathrm{d}s) \le \oint_S V_H(g(s))\varphi_H(\mathrm{d}s),$$
 (A.1)

where V_H is unique up to a positive linear transformation and φ_H is unique. The proof of this claim follows by the proof of the main Theorem in [43]. Notice that in [43] acts are functions mapping S to the set Y of finite support probability measures defined on $\mathcal{P}(X)$, nevertheless, only the fact that Yis a mixture set is used in such proof. Hence, our claim simply follows by taking $Y = \mathbf{M}(\mathcal{U})$.

Let \trianglelefteq_H be the relation induced by \precsim_H on $\mathbf{M}(\mathcal{U})$ through constant acts with asymmetric part \triangleleft_H . The function V_H is such that

- $\mu_1 \leq_H \mu_2 \iff V_H(\mu_1) \leq V_H(\mu_2)$, for every $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$;
- $V_H(\alpha \mu_1 + (1 \alpha)\mu_2) = \alpha V_H(\mu_1) + (1 \alpha)V_H(\mu_2)$, for every $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$ and $\alpha \in [0, 1]$;

moreover, V_H is unique up to a positive linear transformation.

Axiom (A10) implies that, defining for every $B \in \mathcal{U}$

$$v_H(B) = V_H(\delta_B),$$

the function $v_H : \mathcal{U} \to \mathbb{R}$ is bounded by virtue of Lemma 10.5 in [15], and furthermore $V_H(\mu) = \int_{\mathcal{U}} v_H(B)\mu(\mathrm{d}B)$, for every $\mu \in \mathbf{M}(\mathcal{U})$, by virtue of Theorem 10.1 in [15]. Moreover, since $\mathbf{M}(\mathcal{U})$ is the set of all finitely additive probabilities on $\mathcal{P}(\mathcal{U})$, there exist $\mu_*, \mu^* \in \mathcal{P}(\mathcal{U})$ such that

$$V_H(\mu_*) = \inf_{B \in \mathcal{U}} v_H(B)$$
 and $V_H(\mu^*) = \sup_{B \in \mathcal{U}} v_H(B).$

In turn, since for every $\mu \in \mathbf{M}(\mathcal{U})$ we have $V_H(\mu_*) \leq V_H(\mu) \leq V_H(\mu^*)$ and V_H represents \leq_H , for every $f \in \mathcal{F}$ it follows that $\mu_* \leq_H f(s) \leq_H \mu^*$ for all $s \in S$. Hence, by the Corollary in [43] we have that (A.1) holds for all $f, g \in \mathcal{F}$.

We show that axioms (A11) and (A12) imply that, for every $B \in \mathcal{U}$, it holds that

$$\inf_{x \in B} v_H(\{x\}) \le v_H(B) \le \sup_{x \in B} v_H(\{x\}).$$

At this aim, suppose $v_H(B) > \sup_{x \in B} v_H(\{x\}) = \alpha$. By axiom (A12) there exists $x_2 \in X$ such that $\delta_B \leq_H \delta_{\{x_2\}}$ implying that, since V_H represents \leq_H , $v_H(\{x_2\}) \geq v_H(B) > \alpha$. Let $y \in B$ be such that $v_H(\{y\}) \leq \alpha < v_H(B) \leq v_H(\{x_2\})$. Since V_H is affine, there exists $\beta \in [0, 1]$ such that

$$\beta v_H(\{x_2\}) + (1 - \beta) v_H(\{y\}) = \beta V_H(\delta_{\{x_2\}}) + (1 - \beta) V_H(\delta_{\{y\}})$$
$$= V_H(\beta \delta_{\{x_2\}} + (1 - \beta) \delta_{\{y\}}) \in (\alpha, v_H(B)).$$

Moreover, the belief functions $\varphi_{\delta_{\{x_2\}}}, \varphi_{\delta_{\{y\}}}$ induced by $\delta_{\{x_2\}}, \delta_{\{y\}}$ on $\mathcal{P}(X)$ are finitely additive probabilities and the same holds for the belief function φ_{μ} induced on $\mathcal{P}(X)$ by $\mu = \beta \delta_{\{x_2\}} + (1 - \beta) \delta_{\{y\}}$. Since $V_H(\mu) > \alpha =$ $\sup_{x \in B} v_H(\{x\})$ it follows that $V_H(\mu) > V_H(\delta_{\{x\}})$ for every $x \in B$ and axiom (A11) implies

$$\delta_B \leq H \mu$$
,

so, it cannot be $\mu \triangleleft_H \delta_B$ reaching a contradiction since

$$V_H(\mu) < V_H(\delta_B) \Longrightarrow \mu \lhd_H \delta_B.$$

The other inequality is proved analogously.

Hence, for every $B \in \mathcal{U}$, there exists $\alpha_H(B) \in [0, 1]$ such that

$$v_H(B) = \alpha_H(B) \cdot \inf_{x \in B} v_H(\{x\}) + (1 - \alpha_H(B)) \cdot \sup_{x \in B} v_H(\{x\}),$$

where $\alpha_H(B)$ is unique whenever $\inf_{x \in B} v_H(\{x\}) < \sup_{x \in B} v_H(\{x\})$, and is arbi-

trary otherwise. Making B vary we get a function $\alpha_H : \mathcal{U} \to [0, 1]$.

Now, defining $u_H: X \to \mathbb{R}$ setting, for every $x \in X$,

$$u_H(x) = v_H(\{x\}),$$

it follows that

$$V_H(\mu) = \int_{\mathcal{U}} \left(\alpha_H(B) \cdot \inf_{x \in B} u_H(x) + (1 - \alpha_H(B)) \cdot \sup_{x \in B} u_H(x) \right) \mu(\mathrm{d}B),$$

for every $\mu \in \mathbf{M}(\mathcal{U})$. Moreover, u_H is non-constant, bounded and unique up to a positive linear transformation since V_H is.

Axiom (A9) implies that, for every $H, K \in \mathcal{P}(S)^0$, the relations \leq_H and \leq_K on $\mathbf{M}(\mathcal{U})$ coincide, so, up to a positive linear transformation, we can assume $V_H = V_K = V$, which implies $v_H = v_K = v$ and $u_H = u_K = u$, so we can take $\alpha_H = \alpha_K = \alpha$.

For every $f \in \mathcal{F}$ and $H \in \mathcal{P}(S)^0$, define the functional

$$\underline{\Lambda}_{H}(f) = \oint_{S} \left[\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi_{H}(\mathrm{d}s).$$

By the previous discussion we have that, for every $f,g \in \mathcal{F}$ and every $H \in \mathcal{P}(S)^0$, it holds that

$$f \precsim_H g \Longleftrightarrow \underline{\Lambda}_H(f) \le \underline{\Lambda}_H(g).$$

Axioms (A4) and (A5) imply that there exist $x_*, x^* \in X$ such that $x_* <_H^* x^*$. For every $E \in \mathcal{P}(S)$ define the act $\mathbf{1}_E$ as in (2) and, since u is unique up to a positive linear transformation assume $u(x_*) = 0$ and $u(x^*) = 1$. For every $H \in \mathcal{P}(S)^0$ we have that

$$\underline{\Lambda}_{H}(\mathbf{1}_{E}) = \oint_{S} \left[\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) \mathbf{1}_{E}(s) (\mathrm{d}B) \right] \varphi_{H}(\mathrm{d}s) \\ = \oint_{S} \chi_{E}(s) \varphi_{H}(\mathrm{d}s) = \varphi_{H}(E),$$

where χ_E denotes the indicator of E. Furthermore, define the acts $\mathbf{1} = \mathbf{1}_S$ and $\mathbf{0} = \mathbf{1}_{\emptyset}$.

We have that

$$\underline{\Lambda}_{H}(\mathbf{1}_{E}) = \varphi_{H}(E) = \underline{\Lambda}_{H}(\varphi_{H}(E)\mathbf{1} + (1 - \varphi_{H}(E))\mathbf{0})$$

and since $\underline{\Lambda}_H$ represents \preceq_H it follows that

$$\mathbf{1}_E \sim_H \varphi_H(E) \mathbf{1} + (1 - \varphi_H(E)) \mathbf{0}. \tag{A.2}$$

We show that φ_H is a belief function on $\mathcal{P}(S)$. For every $k \geq 2$ and every $E_1, \ldots, E_k \in \mathcal{P}(S)$, for all $\emptyset \neq I \subseteq \{1, \ldots, k\}$ define the acts

$$\overline{\mu_{\{1,\dots,k\}}^{\cup}} = \varphi_H\left(\bigcup_{i=1}^k E_i\right) \mathbf{1} + \left(1 - \varphi_H\left(\bigcup_{i=1}^k E_i\right)\right) \mathbf{0},$$
$$\overline{\mu_I^{\cap}} = \varphi_H\left(\bigcap_{i\in I} E_i\right) \mathbf{1} + \left(1 - \varphi_H\left(\bigcap_{i\in I} E_i\right)\right) \mathbf{0},$$

and let g, h be defined as in axiom (A6). By (A.2) and axiom (A6) we have that, since $\underline{\Lambda}_H$ represents \preceq_H , $\underline{\Lambda}_H(h) \leq \underline{\Lambda}_H(g)$ which is equivalent to

$$\frac{1}{2^{k-1}} \left(\sum_{\substack{\emptyset \neq I \subseteq \{1, \dots, k\} \\ |I| \text{ odd}}} \underline{\Lambda}_H(\overline{\mu_I^{\cap}}) \right) \leq \frac{1}{2^{k-1}} \left(\underline{\Lambda}_H(\overline{\mu_{\{1, \dots, k\}}^{\cup}}) + \sum_{\substack{\emptyset \neq I \subseteq \{1, \dots, k\} \\ |I| \text{ even}}} \underline{\Lambda}_H(\overline{\mu_I^{\cap}}) \right),$$

that is finally equivalent to

$$\varphi_H\left(\bigcup_{i=1}^k E_i\right) \ge \sum_{\emptyset \neq I \subseteq \{1,\dots,k\}} (-1)^{|I|+1} \varphi_H\left(\bigcap_{i \in I} E_i\right),$$

and so φ_H is a belief function.

Define the function $\varphi : \mathcal{P}(S) \times \mathcal{P}(S)^0 \to [0,1]$ setting, for every $E|H \in$ $\mathcal{P}(S) \times \mathcal{P}(S)^0,$

$$\varphi(E|H) = \varphi_H(E).$$

We show that $\varphi(\cdot|\cdot)$ is a full conditional belief function, i.e., it satisfies conditions (i)-(iii') of Definition 1.

Condition (i). For every $H \in \mathcal{P}(S)^0$, since $\mathbf{1}_E(s) = \mathbf{1}_{E \cap H}(s)$, for all $s \in H$, axiom (A7) implies, as $\underline{\Lambda}_H$ represents \preceq_H , that

$$\varphi(E|H) = \varphi_H(E) = \varphi_H(E \cap H) = \varphi(E \cap H|H).$$

Condition (ii). For every $H \in \mathcal{P}(S)^0$, $\varphi(\cdot|H)$ is a belief function on $\mathcal{P}(S)$ since $\varphi_H(\cdot)$ is.

Condition (iii'). For every $A \in \mathcal{P}(S)$ and $B, C \in \mathcal{P}(S)^0$ with $A \subseteq B \subseteq$ C, we need to show that

$$\varphi(A|C) = \varphi_C(A) = \varphi_B(A) \cdot \varphi_C(B) = \varphi(A|B) \cdot \varphi(B|C).$$

Since $\mathbf{1}_B(s) = \mathbf{1}(s)$, for all $s \in B$, axiom (A7) implies $\mathbf{1}_B \sim_B \mathbf{1}$, moreover, from (A.2) it holds

$$\mathbf{1}_A \sim_B \varphi_B(A)\mathbf{1} + (1 - \varphi_B(A))\mathbf{0}.$$

Applying axiom (A3) we get

$$\varphi_B(A)\mathbf{1}_B + (1 - \varphi_B(A))\mathbf{0} \sim_B \varphi_B(A)\mathbf{1} + (1 - \varphi_B(A))\mathbf{0},$$

and axiom (A1) implies

$$\mathbf{1}_A \sim_B \varphi_B(A)\mathbf{1}_B + (1 - \varphi_B(A))\mathbf{0}.$$

.

Since we have

$$\mathbf{1}_{A} = \begin{cases} \delta_{\{x^{*}\}} & \text{if } s \in A, \\ \delta_{\{x_{*}\}} & \text{if } s \in A^{c}, \end{cases} \text{ and}$$
$$\varphi_{B}(A)\mathbf{1}_{B} + (1 - \varphi_{B}(A))\mathbf{0} = \begin{cases} \varphi_{B}(A)\delta_{\{x^{*}\}} + (1 - \varphi_{B}(A))\delta_{\{x_{*}\}} & \text{if } s \in B, \\ \delta_{\{x_{*}\}} & \text{if } s \in B^{c}, \end{cases}$$

1 0

with

$$\begin{array}{lll} \delta_{\{x_*\}} & \trianglelefteq_B & \varphi_B(A)\delta_{\{x^*\}} + (1 - \varphi_B(A))\delta_{\{x_*\}} & \trianglelefteq_B & \delta_{\{x^*\}}, \\ \delta_{\{x_*\}} & \trianglelefteq_C & \varphi_B(A)\delta_{\{x^*\}} + (1 - \varphi_B(A))\delta_{\{x_*\}} & \trianglelefteq_C & \delta_{\{x^*\}}, \end{array}$$

axiom (A8) implies

$$\mathbf{1}_A \sim_C \varphi_B(A) \mathbf{1}_B + (1 - \varphi_B(A)) \mathbf{0}.$$

Moreover, by (A.2) it follows

$$\mathbf{1}_B \sim_C \varphi_C(B) \mathbf{1} + (1 - \varphi_C(B)) \mathbf{0},$$

and applying axiom (A3) we have

$$\varphi_B(A)\mathbf{1}_B + (1 - \varphi_B(A))\mathbf{0} \sim_C \varphi_B(A)[\varphi_C(B)\mathbf{1} + (1 - \varphi_C(B))\mathbf{0}] + (1 - \varphi_B(A))\mathbf{0}$$

= $\varphi_B(A)\varphi_C(B)\mathbf{1} + (1 - \varphi_B(A)\varphi_C(B))\mathbf{0},$

and by axiom (A1) we get

$$\mathbf{1}_A \sim_C \varphi_B(A)\varphi_C(B)\mathbf{1} + (1 - \varphi_B(A)\varphi_C(B))\mathbf{0}.$$

Since $\underline{\Lambda}_C$ represents \preceq_C , and $\underline{\Lambda}_C(\mathbf{1}_A) = \varphi_C(A)$ and $\underline{\Lambda}_C(\varphi_B(A)\varphi_C(B)\mathbf{1} + (1 - \varphi_B(A)\varphi_C(B))\mathbf{0}) = \varphi_B(A)\varphi_C(B)$ it follows

$$\varphi(A|C) = \varphi_C(A) = \varphi_B(A) \cdot \varphi_C(B) = \varphi(A|B) \cdot \varphi(B|C).$$

Moreover, $\varphi(\cdot|\cdot)$ is unique since every φ_H is.

Finally, define the conditional functional $\underline{\Lambda}(\cdot|\cdot)$ setting, for every $f \in \mathcal{F}$ and $H \in \mathcal{P}(S)^0$,

$$\underline{\Lambda}(f|H) = \underline{\Lambda}_{H}(f)$$

$$= \oint_{S} \left[\int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) f(s)(\mathrm{d}B) \right] \varphi(\mathrm{d}s|H).$$

Proof of Proposition 1. For every $H \in \mathcal{P}(S)^0$, by equations (4) and (5), for all $f \in \mathcal{F}$, we have that

$$\underline{\Lambda}(f|H) = \oint_{S} V(f(s)) \varphi(\mathrm{d}s|H),$$

where $V : \mathbf{M}(\mathcal{U}) \to \mathbb{R}$ is defined, for every $\mu \in \mathbf{M}(\mathcal{U})$, as

$$V(\mu) = \int_{\mathcal{U}} \left(\alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x) \right) \mu(\mathrm{d}B).$$

Notice that the boundedness of u implies the boundedness of V, further, for every $H \in \mathcal{P}(S)^0$, it holds that

$$\begin{array}{ll} \mu_1 \trianglelefteq_H \mu_2 & \Longleftrightarrow & V(\mu_1) \le V(\mu_2), \\ x_1 \le^*_H x_2 & \Longleftrightarrow & u(x_1) \le u(x_3), \end{array}$$

for all $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$ and all $x_1, x_2 \in X$.

Since $\mathbf{M}(\mathcal{U})$ is a mixture set and the function V is affine, i.e., it is such that, for every $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$ and $\alpha \in [0, 1]$,

$$V(\alpha \mu_1 + (1 - \alpha)\mu_2) = \alpha V(\mu_1) + (1 - \alpha)V(\mu_2),$$

then necessity of axioms (A1)–(A5) follows by the main Theorem in [43]. As already noticed in the proof of our Theorem 1, in the proof of the quoted theorem we can take $Y = \mathbf{M}(\mathcal{U})$.

We prove necessity of **(A6)**. For every $H \in \mathcal{P}(S)^0$, for every $k \ge 2$ and every $E_1, \ldots, E_k \in \mathcal{P}(S)$, if $x_*, x^* \in X$ are such that $x_* <_H x^*$ and for all $\emptyset \neq I \subseteq \{1, \ldots, k\}$

$$\mathbf{1}_{\bigcup_{i=1}^{k} E_{i}} \sim_{H} \overline{\mu_{\{1,\ldots,k\}}^{\cup}} \text{ and } \mathbf{1}_{\bigcap_{i \in I} E_{i}} \sim_{H} \overline{\mu_{I}^{\cap}}$$

with $\mu_{\{1,\ldots,k\}}^{\cup}, \mu_I^{\cap} \in \mathbf{M}(\mathcal{U})$, consider the acts

$$g = \frac{1}{2^{k-1}} \left(\frac{\overline{\mu_{\{1,\dots,k\}}^{\cup}}}{\mu_{\{1,\dots,k\}}^{\vee}} + \sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\}\\|I| \text{ even}}} \overline{\mu_I^{\cap}} \right) \text{ and } h = \frac{1}{2^{k-1}} \left(\sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\}\\|I| \text{ odd}}} \overline{\mu_I^{\cap}} \right).$$

Since $\underline{\Lambda}(\cdot|H)$ represents \preceq_H , we have that

$$\begin{split} \underline{\Lambda}\left(\mathbf{1}_{\bigcup_{i=1}^{k}E_{i}}\middle|H\right) &= (u(x^{*}) - u(x_{*}))\varphi\left(\bigcup_{i=1}^{k}E_{i}\middle|H\right) + u(x_{*}) = \beta_{\{1,\dots,k\}}^{\cup};\\ \underline{\Lambda}\left(\overline{\mu_{\{1,\dots,k\}}^{\cup}}\middle|H\right) &= V\left(\mu_{\{1,\dots,k\}}^{\cup}\right) = \beta_{\{1,\dots,k\}}^{\cup};\\ \underline{\Lambda}\left(\mathbf{1}_{\bigcap_{i\in I}E_{i}}\middle|H\right) &= (u(x^{*}) - u(x_{*}))\varphi\left(\bigcap_{i\in I}E_{i}\middle|H\right) + u(x_{*}) = \beta_{I}^{\cap};\\ \underline{\Lambda}\left(\overline{\mu_{I}^{\cap}}\middle|H\right) &= V\left(\mu_{I}^{\cap}\right) = \beta_{I}^{\cap}; \end{split}$$

where $\beta_{\{1,\dots,k\}}^{\cup}, \beta_I^{\cap} \in \mathbb{R}$. Further, since f and g are constant acts, by the affinity of V, we have that

$$\begin{split} \underline{\Lambda}(g|H) &= \frac{1}{2^{k-1}} \left(\beta_{\{1,\dots,k\}}^{\cup} + \sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\} \\ |I| \text{ even}}} \beta_I^{\cap} \right), \\ \underline{\Lambda}(h|H) &= \frac{1}{2^{k-1}} \left(\sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\} \\ |I| \text{ odd}}} \beta_I^{\cap} \right). \end{split}$$

Finally, the complete monotonicity of $\varphi(\cdot|H)$ implies that

$$\frac{1}{2^{k-1}} \left(\beta_{\{1,\dots,k\}}^{\cup} + \sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\}\\|I| \text{ even}}} \beta_I^{\cap} \right) \ge \frac{1}{2^{k-1}} \left(\sum_{\substack{\emptyset \neq I \subseteq \{1,\dots,k\}\\|I| \text{ odd}}} \beta_I^{\cap} \right),$$

and since $\underline{\Lambda}(\cdot|H)$ represents \preceq_H , this is equivalent to $h \preceq_H g$.

We prove necessity of (A7). For every $H \in \mathcal{P}(S)^0$, let $f, g \in \mathcal{F}$ be such that f(s) = g(s), for all $s \in H$. Denoting by χ_H the indicator of H, for every $t \in \mathbb{R}$ we have that

$$\begin{split} \varphi(\{(V \circ f) \ge t\}|H) &= \varphi(\{(V \circ f) \ge t\} \cap H|H) \\ &= \varphi(\{(V \circ f)\chi_H \ge t\} \cap H|H) \\ &= \varphi(\{(V \circ f)\chi_H \ge t\}|H) \\ &= \varphi(\{(V \circ g)\chi_H \ge t\}|H) \\ &= \varphi(\{(V \circ g)\chi_H \ge t\} \cap H|H) \\ &= \varphi(\{(V \circ g) \ge t\} \cap H|H) = \varphi(\{(V \circ g) \ge t\}|H). \end{split}$$

Hence, by the definition of the Choquet integral (see, e.g., [10, 25, 42]) we have that $\underline{\Lambda}(f|H) = \underline{\Lambda}(g|H)$, and since $\underline{\Lambda}(\cdot|H)$ represents \preceq_H , this is equivalent to $f \preceq_H g$.

We prove necessity of **(A8)**. For every $A \in \mathcal{P}(S)$ and $B, C \in \mathcal{P}(S)^0$ with $A \subseteq B \subseteq C$, for every $\mu_1, \mu_2, \mu_3 \in \mathbf{M}(\mathcal{U})$ with $\mu_1 \trianglelefteq_B \mu_2 \trianglelefteq_B \mu_3$ and $\mu_1 \trianglelefteq_C \mu_2 \trianglelefteq_C \mu_3$, let $f, g \in \mathcal{F}$ be such that $f_{|A} = \overline{\mu_3}_{|A}, f_{|A^c} = \overline{\mu_1}_{|A^c}, g_{|B} = \overline{\mu_2}_{|B}$ and $g_{|B^c} = \overline{\mu_1}_{|B^c}$. Let $V(\mu_i) = \beta_i \in \mathbb{R}$, for i = 1, 2, 3. We have that

$$\underline{\Lambda}(f|B) = (\beta_3 - \beta_1)\varphi(A|B) + \beta_1,$$

$$\underline{\Lambda}(g|B) = (\beta_2 - \beta_1)\varphi(B|B) + \beta_1 = \beta_2,$$

$$\underline{\Lambda}(f|C) = (\beta_3 - \beta_1)\varphi(A|C) + \beta_1,$$

$$\underline{\Lambda}(g|C) = (\beta_2 - \beta_1)\varphi(B|C) + \beta_1.$$

Since $\underline{\Lambda}(\cdot|B)$ represents \preceq_B , the statement $f \sim_B g$ is equivalent to $\underline{\Lambda}(f|B) = \underline{\Lambda}(g|B)$, that is

$$(\beta_3 - \beta_1)\varphi(A|B) + \beta_1 = \beta_2$$

If the above equation holds then

$$\underline{\Lambda}(g|C) = [(\beta_3 - \beta_1)\varphi(A|B) + \beta_1 - \beta_1]\varphi(B|C) + \beta_1$$

= $(\beta_3 - \beta_1)\varphi(A|B)\varphi(B|C) + \beta_1$
= $(\beta_3 - \beta_1)\varphi(A|C) + \beta_1 = \underline{\Lambda}(f|C),$

and since $\underline{\Lambda}(\cdot|C)$ represents \preceq_C , this is equivalent to $f \sim_C g$.

We prove necessity of **(A9)**. For every $H, K \in \mathcal{P}(S)^0$ and for every $\overline{\mu_1}, \overline{\mu_2} \in \mathcal{F}_{\text{const}}$, since $\underline{\Lambda}(\overline{\mu_i}|H) = \underline{\Lambda}(\overline{\mu_i}|K) = V(\mu_i)$, for i = 1, 2, and $\underline{\Lambda}(\cdot|H)$ and $\underline{\Lambda}(\cdot|K)$ represent \preceq_H and \preceq_K , respectively, we immediately get that

$$\overline{\mu_1} \precsim_H \overline{\mu_2} \Longleftrightarrow \overline{\mu_1} \precsim_K \overline{\mu_2}.$$

We prove necessity of (A10). For $H \in \mathcal{P}(S)^0$, $\mathcal{B} \in \mathcal{P}(\mathcal{U})$, and $\mu_1, \mu_2 \in \mathbf{M}(\mathcal{U})$, we have that

$$\underline{\Lambda}(\overline{\mu_1}|H) = V(\mu_1), \quad \underline{\Lambda}(\overline{\mu_2}|H) = V(\mu_2), \text{ and } \underline{\Lambda}(\overline{\delta_B}|H) = V(\delta_B).$$

Suppose $\mu_1(\mathcal{B}) = 1$ and $\overline{\mu_2} \prec_H \overline{\delta_B}$ for all $B \in \mathcal{B}$. Since $\underline{\Lambda}(\cdot|H)$ represents \preceq_H , the statement $\overline{\mu_2} \prec_H \overline{\delta_B}$ is equivalent to $V(\mu_2) < V(\delta_B)$ for all $B \in \mathcal{B}$. Hence, we get

$$V(\mu_2) \le \inf_{B \in \mathcal{B}} V(\delta_B) \le V(\mu_1),$$

and this is equivalent to $\overline{\mu_2} \preceq_H \overline{\mu_1}$. Analogously, suppose $\mu_1(\mathcal{B}) = 1$ and $\overline{\delta_B} \prec_H \overline{\mu_2}$ for all $B \in \mathcal{B}$. Since $\underline{\Lambda}(\cdot|H)$ represents \preceq_H , the statement $\overline{\delta_B} \prec_H \overline{\mu_2}$ is equivalent to $V(\delta_B) < V(\mu_2)$ for all $B \in \mathcal{B}$. Hence, we get

$$V(\mu_1) \le \sup_{B \in \mathcal{B}} V(\delta_B) \le V(\mu_2),$$

and this is equivalent to $\overline{\mu_1} \preceq_H \overline{\mu_2}$.

We prove necessity of (A11). For $H \in \mathcal{P}(S)^0$, $B \in \mathcal{U}$, and $\mu \in \mathbf{M}(\mathcal{U})$ whose corresponding belief function φ_{μ} on $\mathcal{P}(X)$ is a finitely additive probability, we have that

$$\begin{split} \underline{\Lambda}(\delta_{\{x\}}|H) &= u(x), \\ \underline{\Lambda}(\delta_B|H) &= \alpha(B) \cdot \inf_{x \in B} u(x) + (1 - \alpha(B)) \cdot \sup_{x \in B} u(x), \\ \underline{\Lambda}(\overline{\mu}|H) &= V(\mu). \end{split}$$

Suppose $\overline{\delta_{\{x\}}} \prec_H \overline{\mu}$ for all $x \in B$. Since $\underline{\Lambda}(\cdot|H)$ represents \preceq_H , the statement $\overline{\delta_{\{x\}}} \prec_H \overline{\mu}$ is equivalent to $u(x) < V(\mu)$ for all $x \in B$, which implies

$$\underline{\Lambda}(\delta_B|H) \le \sup_{x \in B} u(x) \le V(\mu) = \underline{\Lambda}(\overline{\mu}|H),$$

that is equivalent to $\overline{\delta_B} \preceq_H \overline{\mu}$. Analogously, suppose $\overline{\mu} \prec_H \overline{\delta_{\{x\}}}$ for all $x \in B$. Since $\underline{\Lambda}(\cdot|H)$ represents \preceq_H , the statement $\overline{\mu} \prec_H \overline{\delta_{\{x\}}}$ is equivalent to $V(\mu) < u(x)$ for all $x \in B$, which implies

$$\underline{\Lambda}(\overline{\mu}|H) = V(\mu) \le \inf_{x \in B} u(x) \le \underline{\Lambda}(\delta_B|H),$$

that is equivalent to $\overline{\mu} \preceq_H \overline{\delta_B}$.

Proof of Theorem 2. The proof follows the same steps of the proof of Theorem 1. We show that, for every $B \in \mathcal{U}$, for the function $v_H : \mathcal{U} \to \mathbb{R}$ derived in the proof of Theorem 1 it holds

$$v_H(B) = \inf_{x \in B} v_H(\{x\}).$$

We already know that $v_H(B) \ge \inf_{x \in B} v_H(\{x\})$, so, it is sufficient to prove only the other inequality. At this aim, consider axiom **(A13)** with $x \in B$ fixed arbitrarily. For $A = \{x\}$ and B there exists $y \in B$ such that $\delta_{\{y\}} \le H \delta_{\{x\}}$, possibly y = x. This implies $\delta_B \le H \delta_{\{x\}}$ and from this it follows $v_H(B) \le v_H(\{x\})$. Since this holds for every $x \in B$ we have that

$$v_H(B) \le \inf_{x \in B} v_H(\{x\}).$$

Now, defining $u_H: X \to \mathbb{R}$ setting, for every $x \in X$,

$$u_H(x) = v_H(\{x\}),$$

it follows that, for every $\mu \in \mathbf{M}(\mathcal{U})$,

$$V_H(\mu) = \int_{\mathcal{U}} \left(\inf_{x \in B} u_H(x) \right) \mu(\mathrm{d}B),$$

that is the function $\alpha_H : \mathcal{U} \to [0, 1]$ is constantly equal to 1.

Proof of Theorem 3. The proof is analogous to the proof of Theorem 2, showing that axiom (A13') implies that the function $v_H : \mathcal{U} \to \mathbb{R}$ derived in the proof of Theorem 1 is such that

$$v_H(A) = \sup_{x \in A} v_H(\{x\}).$$

We already know that $v_H(A) \leq \sup_{x \in A} v_H(\{x\})$, so, it is sufficient to prove only the other inequality. At this aim, consider axiom **(A13')** with $y \in A$ fixed arbitrarily. For $B = \{y\}$ and A there exists $x \in A$ such that $\delta_{\{y\}} \leq_H \delta_{\{x\}}$, possibly x = y. This implies $\delta_{\{y\}} \leq_H \delta_A$ and from this it follows $v_H(\{y\}) \leq$ $v_H(A)$. Since this holds for every $y \in A$ we have that

$$v_H(A) \ge \sup_{y \in A} v_H(\{y\}).$$

Now, defining $u_H: X \to \mathbb{R}$ setting, for every $x \in X$,

$$u_H(x) = v_H(\{x\}),$$

it follows that, for every $\mu \in \mathbf{M}(\mathcal{U})$,

$$V_H(\mu) = \int_{\mathcal{U}} \left(\sup_{x \in B} u_H(x) \right) \mu(\mathrm{d}B),$$

that is the function $\alpha_H : \mathcal{U} \to [0, 1]$ is constantly equal to 0.

Proof of Proposition 2. Necessity of axioms (A1)–(A11), both for a representation through $\underline{\Phi}(\cdot|\cdot)$ and $\underline{\Psi}(\cdot|\cdot)$, comes from Proposition 1. To prove necessity of (A13) for the representation through $\underline{\Phi}(\cdot|\cdot)$, let $A, B \in \mathcal{U}$ be such that for all $x \in A$, there exists $y \in B$ such that $\overline{\delta}_{\{y\}} \preceq_H \overline{\delta}_{\{x\}}$. We have that

$$\underline{\Phi}(\overline{\delta_{\{y\}}}|H) = u(y), \quad \underline{\Phi}(\overline{\delta_{\{x\}}}|H) = u(x),$$

$$\underline{\Phi}(\overline{\delta_A}|H) = \inf_{x \in A} u(x), \quad \underline{\Phi}(\overline{\delta_B}|H) = \inf_{y \in B} u(y),$$

and since $\underline{\Phi}(\cdot|H)$ represents \leq_H , we get that for all $x \in A$, there exists $y \in B$ such the $u(y) \leq u(x)$, therefore

$$\inf_{y \in B} u(y) \le \inf_{x \in A} u(x),$$

which is equivalent to $\overline{\delta_B} \preceq_H \overline{\delta_A}$. Analogously, to prove necessity of **(A13')** for the representation through $\underline{\Psi}(\cdot|\cdot)$, let $A, B \in \mathcal{U}$ be such that for all $y \in B$, there exists $x \in A$ such that $\overline{\delta_{\{y\}}} \preceq_H \overline{\delta_{\{x\}}}$. We have that

$$\underline{\Psi}(\delta_{\{y\}}|H) = u(y), \quad \underline{\Psi}(\delta_{\{x\}}|H) = u(x),$$

$$\underline{\Psi}(\overline{\delta_A}|H) = \sup_{x \in A} u(x), \quad \underline{\Psi}(\overline{\delta_B}|H) = \sup_{y \in B} u(y),$$

and since $\underline{\Psi}(\cdot|H)$ represents \leq_H , we get that for all $y \in B$, there exists $x \in A$ such the $u(y) \leq u(x)$, therefore

$$\sup_{y \in B} u(y) \le \sup_{x \in A} u(x),$$

which is equivalent to $\overline{\delta_B} \preceq_H \overline{\delta_A}$.

Acknowledgements

The authors are members of the GNAMPA-INdAM research group. The first author was supported by Università degli Studi di Perugia, Fondo Ricerca di Base 2019, project "Modelli per le decisioni economiche e finanziarie in condizioni di ambiguità ed imprecisione". The second author was supported by "La Sapienza" Università di Roma, Bandi 2020, "Modelli decisionali in condizioni di incertezza e ambiguità" (RP120172B81E33DD).

References

- D.S. Ahn. Ambiguity without a state space. The Review of Economic Studies, 75(1):3–28, 2008.
- [2] F.J. Anscombe and R.J. Aumann. A definition of subjective probability. The Annals of Mathematical Statistics, 34(1):199–205, 1963.
- [3] K.P.S. Bhaskara Rao and M. Bhaskara Rao. Theory of Charges: A Study of Finitely Additive Measures. Academic Press, 1983.
- [4] A. Chateauneuf and J.-Y. Jaffray. Some characterizations of lower probabilities and other monotone capacities through the use of Möbius inversion. *Mathematical Social Sciences*, 17(3):263–283, 1989.
- [5] A. Chateauneuf, R. Kast, and A. Lapied. Conditioning Capacities and Choquet Integrals: The Role of Comonotony. *Theory and Decision*, 51(2):367–386, 2001.
- [6] G. Coletti, D. Petturiti, and B. Vantaggi. Conditional belief functions as lower envelopes of conditional probabilities in a finite setting. *Information Sciences*, 339:64–84, 2016.

- [7] G. Coletti and R. Scozzafava. From Conditional Events to Conditional Measures: A New Axiomatic Approach. Annals of Mathematics and Artificial Intelligence, 32(1):373–392, 2001.
- [8] G. Coletti and B. Vantaggi. A view on conditional measures through local representability of binary relations. *International Journal of Approximate Reasoning*, 47(3):268–283, 2008.
- [9] A.P. Dempster. Upper and Lower Probabilities Induced by a Multivalued Mapping. Annals of Mathematical Statistics, 38(2):325–339, 1967.
- [10] D. Denneberg. Non-Additive Measure and Integral. Kluwer Academic Publisher, 1994.
- [11] L.E. Dubins. Finitely additive conditional probabilities, conglomerability and disintegrations. The Annals of Probability, 3(1):89–99, 1975.
- [12] D. Ellsberg. Risk, Ambiguity, and the Savage Axioms. The Quarterly Journal of Economics, 75(4):643–669, 1961.
- [13] J. Etner, M. Jeleva, and J.-M. Tallon. Decision Theory under Ambiguity. Journal of Economic Surveys, 26(2):234–270, 2012.
- [14] R. Fagin and J.Y. Halpern. Uncertainty, belief, and probability. Computational Intelligence, 7(3):160–173, 1991.
- [15] P. Fishburn. Utility Theory for Decision Making. John Wiley and Sons, New York, 1970.
- [16] P.C. Fishburn. The foundations of expected utility. Springer Science, Dordrecht, 1982.
- [17] P. Ghirardato. Coping with ignorance: unforeseen contingencies and non-additive uncertainty. *Economic Theory*, 17:247–276, 2001.
- [18] P. Ghirardato, F. Maccheroni, and M. Marinacci. Revealed Ambiguity and Its Consequences: Updating. In M. Abdellaoui and J.D. Hey, editors, Advances in Decision Making Under Risk and Uncertainty, pages 3–18, Berlin, Heidelberg, 2008. Springer Berlin Heidelberg.
- [19] P.H. Giang. Subjective foundation of possibility theory: Anscombe-Aumann approach. *Information Sciences*, 370-371:368–384, 2016.
- [20] I. Gilboa. *Theory of Decision under Uncertainty*. Cambridge University Press, 2009.

- [21] I. Gilboa and M. Marinacci. Ambiguity and the Bayesian Paradigm, chapter 21. Springer International Publishing Switzerland 2016.
- [22] I. Gilboa and D. Schmeidler. Maxmin expected utility with non-unique prior. Journal of Mathematical Economics, 18(2):141–153, 1989.
- [23] I. Gilboa and D. Schmeidler. Updating ambiguous beliefs. Journal of Econonomic Theory, 59:33–49, 1993.
- [24] I. Gilboa and D. Schmeidler. Canonical representation of set functions. Mathematics of Operations Research, 20(1):197–212, 1995.
- [25] M. Grabisch. Set Functions, Games and Capacities in Decision Making. Springer, 2016.
- [26] F. Gul and W. Pesendorfer. Expected Uncertain Utility Theory. Econometrica, 82(1):1–39, 2014.
- [27] E. Hanany and P. Klibanoff. Updating preferences with multiple priors. *Theoretical Economics*, 2:261–298, 2007.
- [28] M. Horie. A unified representation of conditioning rules for convex capacities. *Economics Bulletin*, 4(19):1–6, 2006.
- [29] L. Hurwicz. Optimality Criteria for Decision Making under Ignoranc. Technical Report 370, Cowles Commission Papers, 1951.
- [30] J.-Y. Jaffray. Linear utility theory for belief functions. Operations Research Letters, 8(2):107–112, 1989.
- [31] J.-Y. Jaffray. Bayesian updating and belief functions. IEEE Transactions on Systems, Man, and Cybernetics, 22(2):1144–1552, 1992.
- [32] R. Kast, A. Lapied, and P. Toquebeuf. Updating Choquet capacities: a general framework. *Economics Bulletin*, 32(2):1495–1503, 2012.
- [33] D.M. Kreps. A representation theorem for "preference for flexibility". *Econometrica*, 47(3):56–577, 1979.
- [34] D.M. Kreps and R. Wilson. Sequential equilibria. Econometrica, 50(4):863–894, 1982.
- [35] M. Marinacci. Decomposition and representation of coalitional games. Mathematics of Operations Research, 21(4):1000–1015, 1996.

- [36] R.B. Myerson. Game Theory: Analysis of Conflict. Harvard University Press, 1991.
- [37] W. Olszewski. Preferences over sets of lotteries. The Review of Economic Studies, 74(2):567–595, 2007.
- [38] C. Pacheco Pires. A Rule For Updating Ambiguous Beliefs. Theory and Decision, 53(2):137–152, 2002.
- [39] D. Petturiti and B. Vantaggi. Conditional submodular Choquet expected values and conditional coherent risk measures. *International Journal of Approximate Reasoning*, 113:14–38, 2019.
- [40] D. Petturiti and B. Vantaggi. Modeling agent's conditional preferences under objective ambiguity in Dempster-Shafer theory. *International Journal of Approximate Reasoning*, 119:151–176, 2020.
- [41] L.J. Savage. The foundations of statistics. Wiley, New York, 1954.
- [42] D. Schmeidler. Integral representation without additivity. *Proceedings* of the American Mathematical Society, 97(2):255–261, 1986.
- [43] D. Schmeidler. Subjective probability and expected utility without additivity. *Econometrica*, 57(3):571–587, 1989.
- [44] G. Shafer. A Mathematical Theory of Evidence. Princeton University Press, Princeton, NJ, 1976.
- [45] G. Shafer. A theory of statistical evidence. In W.L. Harper and C.A. Hooker, editors, Foundations of Probability Theory, Statistical Inference, and Statistical Theories of Science, volume 6b of The University of Western Ontario Series in Philosophy of Science, pages 365–436. Springer Netherlands, 1976.
- [46] P. Suppes and M. Zanotti. On using random relations to generate upper and lower probabilities. Synthese, 36(4):427–440, 1977.
- [47] J.-M. Tallon and J.-C. Vergnaud. Knowledge, Beliefs and Economics, chapter Beliefs and dynamic consistency, pages 137–154. Edward Elgar Publishing, 2006.
- [48] M.-L. Vierø. Exactly what happens after the Anscombe–Aumann race? Economic Theory, 41(2):175–212, 2009.

- [49] P.P. Wakker. Nonexpected utility as aversion of information. Journal of Behavioral Decision Making, 1:169–175, 1998.
- [50] P. Walley. Coherent lower (and upper) probabilities. Technical report, Department of Statistics, University of Warwick, 1981.